Management Quality and Antitakeover Provisions

Thomas J. Chemmanur Boston College Imants Paeglis Concordia University Karen Simonyan Suffolk University

Abstract

We present the first empirical analysis of the relationship between a firm's management quality and the prevalence of antitakeover provisions in its corporate charter and their influence on initial public offering (IPO) valuation and post-IPO performance. We test the implications of the managerial entrenchment hypothesis, which implies that antitakeover provisions serve only to enhance the control benefits of incumbent management, and the long-term value creation hypothesis, which implies that such provisions can enhance value in the hands of higher quality management. We find that, first, firms with higher quality management and greater growth options are associated with a greater number of antitakeover provisions. Second, firms with higher management quality and a greater number of antitakeover provisions outperform other firms in the sample in terms of post-IPO operating and stock return performance and obtain higher IPO valuations. Our findings reject the managerial entrenchment hypothesis and support the long-term value creation hypothesis.

1. Introduction

Why do firms adopt various antitakeover provisions (ATPs) in their corporate charters and bylaws? How do the presence and strength of such provisions affect the future stock return and operating performance of firms? The answers to these questions are controversial, although a number of papers have attempted

For helpful comments and discussions, we thank Shan He, Gang Hu, Yawen Jiao, Ayla Kayhan, Ji-Chai Lin, Debarshi Nandy, Wei-Ling Song, and Tong Yu as well as seminar participants at Boston College, Suffolk University, Louisiana State University, and University of Rhode Island and conference participants at the 2009 European Finance Association meetings and the 2007 Financial Management Association meetings. Special thanks to an anonymous referee for several helpful comments and suggestions. We alone are responsible for any errors or omissions.

[Journal of Law and Economics, vol. 54 (August 2011)]
© 2011 by The University of Chicago. All rights reserved. 0022-2186/2011/5403-0021\$10.00

to answer these questions in various contexts (Agrawal and Mandelker 1990; Agrawal and Knoeber 1996). The objective of this paper is to shed new light on these questions by analyzing, for the first time in the literature, the relationship between the quality and reputation of a firm's management and the prevalence of ATPs in its corporate charter and the relationship between these two variables and the firm's valuation and operating and stock return performance using a sample of firms going public. Initial public offerings (IPOs) of equity are a particularly appropriate context in which to study these relationships. Many provisions in a firm's corporate charter are decided on at the time of going public, which allows us to study contemporaneously the relationship between the number of ATPs in a firm's corporate charter and the quality of the management team that put these provisions in place. In order to analyze that relationship, we test the implications of two alternative theories regarding the relationship between the quality of a firm's management and the prevalence of ATPs and regarding the relationship among management quality, ATPs, and IPO valuation and post-IPO performance.

The first theory we test is the managerial entrenchment hypothesis. This hypothesis argues that ATPs reduce shareholder value since they entrench managers by reducing the probability of takeovers by rival management teams and thus insulate managers from the discipline imposed by the market for corporate control. Such managerial entrenchment may allow a firm's management to exert less effort in running the firm and allow managers to extract more control benefits. This implies that firms with lower management quality are more likely to have stronger ATPs in their corporate charters when going public. Further, this theory implies that regardless of the ability of the management team, stronger ATPs lead to poorer post-IPO firm performance and lower IPO valuation since they reduce the disciplining effect of the market for corporate control on firm management.

While there is some evidence (Field and Karpoff 2002) that ATPs play a role in entrenching firms' managers, other evidence indicates that there is more to the use of ATPs in corporate charters than a blatant attempt to entrench current management at the expense of shareholders. For example, the empirical evidence indicates that the use of ATPs in IPOs has increased rather than decreased over time.² Further, in contrast to the view that ATPs are value reducing, Field and Karpoff (2002) do not find any evidence that ATPs at the time of IPO contribute to poor post-IPO operating performance. In fact, their evidence indicates that post-IPO operating performance in the years immediately after the IPO is poorer for firms without takeover defenses than for firms with defenses. In summary, the management entrenchment hypothesis merits further empirical examination.

¹ In contrast, the relationship between management quality and antitakeover provisions (ATPs) in a firm's corporate charter is harder to analyze for more seasoned firms because many of these provisions may have been inherited from previous management teams.

² For example, Bebchuk (2003) points out that 82 percent of firms going public in 2002 made use of staggered boards, while only 35 percent of firms going public during 1988–92 did so.

We propose to accomplish this by studying the relationship between the quality of a firm's management and the prevalence of ATPs in its corporate charter.

The second hypothesis we test is the long-term value creation hypothesis developed by Chemmanur and Jiao (2012). Their theoretical analysis demonstrates that, in an environment of asymmetric information about management quality, dual-class share structures and other ATPs may be value enhancing in the hands of higher quality (more talented) managers. Such ATPs allow managers to create superior value for the firm by investing in risky long-term projects without fear of losing control to inferior rivals in a control contest (for example, in a situation in which the firm's project is in temporary difficulties). Antitakeover provisions will be value destroying in the hands of lower quality managers since they can use such provisions to enjoy the benefits of control without being able to create any superior long-term value.

The long-term value creation hypothesis implies that firms with more reputable managers will be more likely to include stronger ATPs in their corporate charters, since the ability to create long-term value will dominate any reduction in the IPO share price imposed by the equity market. Next, among firms with higher management quality, those with larger growth options will have stronger ATPs since there is a greater opportunity for value creation in such firms. Further, for firms with higher quality managers, having stronger ATPs will result in better post-IPO performance and higher IPO valuation. Thus, we divide a sample of IPO firms into four groups: firms with high or low management quality, with each category subdivided into those with a greater or smaller number of ATPs. The prediction is that firms in the group with high-quality managers and a greater number of ATPs will have, on average, better post-IPO operating and stock return performance and higher IPO valuation than firms in the remaining three groups.³

We test the implications of the two theories using a sample of firms going public between 1993 and 2000 and making use of measures of management quality developed by Chemmanur and Paeglis (2005). Data on management quality were hand collected from IPO prospectuses. Data on ATPs in the charters of firms going public were also hand collected from IPO prospectuses. We study 19 ATPs at the firm level as described in Appendix A.

Our empirical results are as follows. First, firms with higher management quality have, on average, a greater number of ATPs in their corporate charters.

³ The existing literature also advances a shareholder interest argument for the adoption of ATPs in corporate charters. Under the shareholder interest hypothesis, ATPs are adopted to increase the bargaining power of management when dealing with corporate acquisitions, thus yielding higher takeover premia for selling shareholders: see Comment and Schwert (1995), who document that ATPs are associated with higher takeover premia, and Linn and McConnell (1983), who document a positive announcement effect on firms' equity on the adoption of ATPs. While, like the long-term value creation hypothesis, the shareholder interest hypothesis also predicts that the adoption of ATPs increases shareholder wealth, it does not have any predictions for the relationship between the quality of a firm's management and the strength of the ATPs in its corporate charter, which is the primary focus of this paper.

Second, on average, firms with higher management quality and larger growth options have significantly more ATPs in their charters compared with all other firms. We also find that firms with higher management quality and a greater number of ATPs have, on average, better long-term post-IPO operating and stock return performance and higher IPO valuation compared with all other firms

These results contradict the idea that the role of ATPs in IPO charters is solely to entrench firm management, for several reasons. First, if this were the case, one would expect firms with lower quality managers also to have a significant number of ATPs: in other words, the number of ATPs would not be increasing in management quality. Second, our finding that the number of ATPs is greater for firms with higher management quality and larger growth options indicates that ATPs may help higher quality managers undertake investments in long-term projects without fear of loss of control, if such projects are in temporary difficulties, by providing them some insulation from the takeover market. Finally, our finding that firms with higher quality management and a greater number of ATPs have better post-IPO performance and higher IPO valuation indicates that ATPs may be an efficient mechanism that enables higher quality managers to create greater shareholder value by investing in long-term projects without being subject to the short-term pressures generated by the market for corporate control.

Our paper is related to several strands in the literature. The first strand is the literature analyzing the role of ATPs in the context of various corporate events. Examples include Borokhovich, Brunarski, and Parrino (1997), who document that after a firm adopts ATPs, takeovers become less likely and managers tend to increase their own pay; Masulis, Wang, and Xie (2007), who document that acquirers with more ATPs have lower abnormal returns around the time of acquisition announcements; Garvey and Hanka (1999), who document that firms reduce debt levels after adopting ATPs; Hartzell, Kallberg, and Liu (2008), who use a sample of real estate investment trusts (REITs) to study the relationship between the corporate governance structure of a firm and its valuation at the time of IPO; and Field and Karpoff (2002) and Daines and Klausner (2001), who study ATPs in the context of firms going public. It is important to note that, while some of these papers investigate the relationship between ATPs in the corporate charters of firms going public and their subsequent performance, ours is the first paper that studies the role of management quality in this relationship.4

The second strand is the literature relating the prevalence of ATPs in a firm's corporate charter with shareholder value. A prominent example is Gompers,

⁴ Our paper is also related to the broader literature that links variation in state antitakeover statutes and the provisions in corporate debt covenants (see, for example, Qi and Wald 2008). It is also indirectly related to the literature analyzing the relationship between corporate governance mechanisms characterizing a firm and the incidence of earnings management or earnings restatement (see, for example, Agrawal and Chadha 2005).

Ishii, and Metrick (2003), who find that firms with a greater number of ATPs have lower stock returns. Core, Guay, and Rusticus (2006), however, question the finding, arguing that there is no conclusive evidence that having a greater number of ATPs causes poorer stock returns. In another related paper, Gompers, Ishii, and Metrick (2004) study the valuation of dual-class firms (as measured by Tobin's *q*) and document that firm value is increasing in firm insiders' cash flow ownership but decreasing in their voting ownership. They recognize, however, that management quality may be an omitted variable in their analysis. "It is . . . possible that these results are driven by some outside factor: e.g., valuation is driven by some measure of 'management quality,' and management quality in turn drives the particular form of dual-class structure adopted across firms' (Gompers, Ishii, and Metrick 2004, p. 20). To the extent that dual-class share structures are one of the several ATPs that a firm may include in its corporate charter, our analysis can be thought of as addressing their concern regarding the omission of management quality from existing studies. ^{5,6}

The third strand is the newly emerging literature on the relationship between the management quality of a firm and various aspects of its IPO (Chemmanur and Paeglis 2005) or other financial policies (Bertrand and Schoar 2003). Unlike the present paper, which studies the relationship between management quality and the prevalence and effects of ATPs in a firm's IPO, the focus of Chemmanur and Paeglis (2005) is on the effects of management quality on IPO characteristics such as underpricing, underwriting spread, and other costs of going public; that paper therefore does not address any of the issues that we study here.⁷

By incorporating management quality in our analysis of the relationship between ATPs and corporate performance, our paper complements the important insights provided by the existing literature on why firms adopt ATPs in their corporate charters. Instead of merely entrenching firm managers and thus always reducing shareholder value, our analysis indicates that ATPs play a more nuanced role in affecting shareholder value: while having a greater number of ATPs may indeed destroy shareholder value in the hands of lower quality managers (possibly by entrenching them more strongly), we show that ATPs enhance shareholder value in the hands of higher quality managers. Thus, we provide a more complete picture of the relationship between ATPs and shareholder value. In particular, our analysis greatly enhances the understanding of how firms choose various ATPs to include in their corporate charters when they go public. Further, our

⁵ There are also a number of other papers studying the rationale for and valuation of dual-class voting structures in initial public offerings (IPOs). See, for example, Smart and Zutter (2003).

⁶ There are also a number of event studies of the adoption of ATPs. See, for example, DeAngelo and Rice (1983), Jarrell and Poulsen (1987), Karpoff and Malatesta (1989), and Bhagat and Jefferis (1991).

⁷ The present paper is also related to the broad theoretical and empirical literature on IPOs and the decision to go public. See, for example, Allen and Faulhaber (1989), Chemmanur (1993), and Welch (1989) on IPO underpricing and Chemmanur and Fulghieri (1999) on the decision to go public.

findings provide a rationale for the fact that the use of ATPs in IPOs has increased rather than decreased over time.

The rest of this paper is organized as follows. Section 2 summarizes the relevant theory and develops the hypotheses. Section 3 describes our data and sample selection procedure. Section 4 develops our measures of management quality and reputation and of firm quality and governance (used as control variables in our analysis). Section 5 presents our empirical tests and results. Section 6 concludes.

2. Theory and Hypotheses

There are two broad sets of theories that have implications for why firms adopt ATPs and for IPO valuation and post-IPO performance. The first set of theories can be thought of as emerging from the seminal works of Grossman and Hart (1988) and Harris and Raviv (1988, 1989), which imply that dual-class voting structures and other ATPs are inefficient. Their models consider a setting in which the incumbent management of a firm obtains not only cash flow or security benefits (arising from her equity ownership in the firm) but also private benefits from being in control; outside shareholders receive only security benefits. The models conclude that dual-class voting structures and other ATPs are value reducing since they reduce the chance of takeovers by rival management teams who can increase the cash flows to current shareholders by managing the firm better than the incumbent does. Thus, under these theories, ATPs are inefficient, and the only role of such provisions is to entrench existing management and reduce its chance of losing its benefits of control. We refer to this hypothesis as the managerial entrenchment hypothesis of ATPs.

In contrast, Chemmanur and Jiao (2012) consider a setting in which the incumbent management of a private firm wishes to sell equity to outsiders in an IPO to raise external financing for its project. The incumbent obtains both security benefits (from the equity she owns in the firm) and private benefits of control. The firm can adopt one of two projects: a long-term project or a short-term project. A long-term project is intrinsically more valuable than a short-term project and therefore maximizes long-term value. However, adopting it may cause the firm's equity to be undervalued in the short term, since it may show fewer signs of success in the short run than will a short-term project (in other words, a long-term project takes a longer time to resolve outsiders' uncertainty about a project's success or failure). Thus, the incumbent has a greater chance of losing control to potential rivals (even those less able than she) if she adopts the long-term project and if outside investors believe that the firm's project is not progressing well in the short term. Outside investors may vote for the rival in a control contest and replace the incumbent if the latter does not

⁸ See also Cary (1969) and Williamson (1975), who made earlier, more informal, arguments that ATPs act primarily to entrench incumbent management.

hold enough voting power on her own account to defeat such a rival. The incumbent may be talented or untalented: talented managers have a lower cost of exerting effort and a comparative advantage in implementing projects than do untalented managers. The incumbent's talent is private information: outsiders observe only a prior probability that she is talented (that is, her reputation or perceived management quality). In this situation, the incumbent makes a joint decision about the voting structure and other ATPs in the corporate charter for her firm's IPO, the kind of project to adopt (long-term or short-term), and the extent of effort to exert in implementing this project.

The equilibrium in Chemmanur and Jiao (2012) is driven by the choice made by a truly talented incumbent (since an untalented incumbent would mimic such choices in order to not reveal her true type to the equity market). The choice of a talented incumbent between adopting stronger versus weaker ATPs depends on three effects. First, the insulation from the takeover market provided by stronger ATPs would allow the incumbent to create more value by implementing a long-term rather than a short-term project. Second, such insulation also allows untalented incumbents to slack off by not exerting effort, thus dissipating value without any fear of losing control to potential rivals. Since equity markets cannot perfectly distinguish between talented and untalented incumbents, this loss-of-discipline effect is also reflected in the talented incumbent's firm's IPO share price if she adopts stronger ATPs (and favors her adopting weaker ATPs instead). Third, regardless of the kind of project adopted, there is a greater chance for incumbent management to maintain control under a corporate charter with stronger ATPs.

Chemmanur and Jiao (2012) show that when the reputation of incumbent management is high enough and the firm has greater opportunities for long-term value creation, in equilibrium, management would adopt an IPO charter with stronger ATPs. This is because the long-term value creation effect dominates any reduction in IPO share value arising from the loss of discipline effect, and such stronger ATPs will in fact be value enhancing for shareholders. We refer to this hypothesis of ATPs as the long-term value creation hypothesis.

The long-term value creation hypothesis leads to two testable predictions regarding the prevalence of ATPs in IPO firms' corporate charters. The first prediction is that, on average, higher quality managers will adopt corporate charters with a greater number of ATPs, which generates the first hypothesis that we test.

Hypothesis 1. Firms with higher quality management will be associated with a greater number of ATPs.

⁹ Stein (1988) has a model of corporate myopia in which takeover pressure under asymmetric information may make managers invest in short-term rather than in (higher value) long-term projects. However, unlike in Chemmanur and Jiao (2012), there is no role for management quality in the Stein (1988) analysis, so it does not have implications for the relationship between management quality and ATPs.

Notice that this is in direct opposition to the prediction of the managerial entrenchment hypothesis. While models of managerial entrenchment do not incorporate different levels of management quality, allowing for such variation in quality would imply that firms with less able management teams are more likely to have a greater number of ATPs under the managerial entrenchment hypothesis. This is because higher quality managers will be more able to resist future takeover attempts based on attracting votes from outside shareholders in a control contest, so lower quality managers will benefit more from (and therefore will adopt) a greater number of ATPs in their corporate charter.

The second prediction of the long-term value creation hypothesis is that among the firms with higher management quality, those with greater opportunity for long-term value creation (that is, larger growth options) are more likely to adopt a greater number of ATPs in their corporate charters. We divide the IPO sample into four groups based on management quality and growth options: higher management quality with larger growth options (group 1), higher management quality with smaller growth options (group 2), lower management quality with larger growth options (group 3), and lower management quality with smaller growth options (group 4). The prediction is that firms in group 1 will have the greatest number of ATPs (since they have the greatest opportunity for value creation) and that firms in group 4 will have the least number of ATPs (since they have the least opportunity for long-term value creation), which generates the second hypothesis that we test.

Hypothesis 2. Firms with higher quality management and larger growth options will be associated with the greatest number of ATPs, and firms with lower quality management and smaller growth options will be associated with the fewest ATPs.

The managerial entrenchment hypothesis predicts no direct relationship between the extent of growth options available to a firm and the strength of ATPs in its corporate charter.

The long-term value creation hypothesis also has predictions for the relationship among management quality, ATPs, and the post-IPO performance and IPO valuation of firms going public. Since under the long-term value creation hypothesis, having a greater number of ATPs is, in fact, value enhancing for firms with higher quality managers, the prediction is that there will be a positive relationship between the number of ATPs and post-IPO operating performance for such firms. We divide our IPO sample into four categories based on management quality as well as ATPs: higher management quality with stronger ATPs, higher management quality with weaker ATPs, lower management quality with stronger ATPs, and lower management quality with weaker ATPs. We predict that the post-IPO operating performance of firms in the category with higher management quality and stronger ATPs will, on average, be significantly better

than that of firms in the remaining three categories. ¹⁰ Further, if investors anticipate the better operating performance of firms with higher management quality and stronger ATPs, the IPO valuation of this group will be higher than that of firms in the remaining three categories. Finally, since long-term post-IPO stock returns generally move together with post-IPO operating performance, the long-term value creation hypothesis also implies that firms in the first category would, on average, outperform those in the remaining three categories in terms of long-term post-IPO stock returns. ¹¹ Thus, we test the following three hypotheses:

Hypothesis 3. Post-IPO operating performance of firms with higher quality management and a greater number of ATPs will be better than that of firms in the rest of the sample.

Hypothesis 4. The IPO valuation of firms with higher quality management and a greater number of ATPs will be higher than that of firms in the rest of the sample.

Hypothesis 5. Long-term post-IPO stock return performance of firms with higher quality management and a greater number of ATPs will be better than that of firms in the rest of the sample.

In contrast, the predictions of the managerial entrenchment hypothesis regarding the relationship between management quality and ATPs on the one hand and IPO valuation, post-IPO operating, and stock return performance on the

¹⁰ Note that, in the setting of Chemmanur and Jiao (2012), all four combinations will arise in equilibrium. This is because the number of ATPs included in a firm's corporate charter depends on the trade-off between the short-term and the long-term effects on top management's objective of including them in the corporate charter: while including a larger number of ATPs will lead to a lower short-term IPO share price (because of the loss-of-discipline effect discussed earlier), it will lead to a better long-term operating (and stock return) performance. Thus, we will observe both high-management-quality firms with stronger ATPs and high-management-quality firms with weaker ATPs. Further, since the equilibrium in Chemmanur and Jiao (2012) is a pooling equilibrium in which high- and low-management-quality firms pool together in the IPO market, we will also find low-management-quality firms with stronger ATPs and low-management-quality firms with weaker ATPs in their corporate charters.

¹¹ If outside investors are fully rational and the stock market is completely efficient, one should not observe any differences in the postissue long-run stock return performance of IPO firms with higher management quality and stronger ATPs versus the rest of the sample. If higher management quality and stronger ATPs increase the likelihood of long-term value creation, such information will be reflected in IPO offer price on the issue date: in other words, there will be no differences in the long-run returns measured subsequent to the issue date. If, however, investors are only boundedly rational, so this information is not fully reflected in the IPO offer price but is incorporated only over a longer period, then one would expect better long-run stock return performance from firms with higher management quality and stronger ATPs. Note that all long-run stock return studies around corporate events require the assumption of bounded rationality or limited market efficiency similar to the one we make here. One may consider this to be a strong assumption, but, given the large empirical literature documenting the postevent drift following earnings announcements and many other corporate events (see, for example, Foster, Olsen, and Shevlin 1984; Bernard and Thomas 1989), one has to at least consider the possibility that the information revealed by many corporate actions is not always instantaneously reflected in the stock price.

other are as follows. Given that it may lose control of its firm through a takeover, incumbent management will work harder to manage the firm and attract votes from outside shareholders if the probability of a successful takeover is greater. (See Chemmanur and Yan [2004] for a model that includes effort choice by incumbent management and incorporating the disciplinary effect of takeovers.) Since stronger ATPs reduce the chance of such credible takeover attempts from succeeding (that is, they reduce the threat of takeovers), the managerial entrenchment hypothesis implies that, regardless of management quality, firms with stronger ATPs will have poorer post-IPO operating performance, since incumbent management (whether of high or low quality) is likely to work less hard in such firms. Further, if investors rationally anticipate that firms with stronger ATPs will have poorer post-IPO operating performance, the managerial entrenchment hypothesis implies that such firms will have lower IPO valuations as well. Finally, since long-term stock returns generally move hand in hand with operating performance, the managerial entrenchment hypothesis also implies a negative relationship between the prevalence of ATPs in a firm's IPO corporate charter and post-IPO stock returns.

3. Data and Sample Selection

The list of IPOs of common equity between 1993 and 2000 comes from the Security Data Company (SDC)/Platinum Global New Issues database. After elimination of REITs, closed-end funds, unit offerings, equity carve outs, financial firms (all firms with Security Industry Classification [SIC] codes between 6000 and 6999), foreign companies, previous leveraged buyouts, roll ups, firms not found in the Center for Research in Security Prices (CRSP) and/or Compustat Industrial Annual databases, and firms for which CRSP and SDC show different first dates of trading, we are left with 2,644 firms in our sample.¹²

In order to isolate the effects of management quality on IPO performance and valuation, and to remove any confounding effects arising from the presence of venture capitalists (VCs) or institutional investors as firm backers, we confine our study to non-VC-backed firms and those with no institutional investors prior to the IPO. In addition to the direct effect of VCs on IPO performance and valuation (through VC certification), VCs can affect the management quality of IPO firms either by selecting managers or by performing various roles (for example, legal representation, selection of underwriters) that would be performed by management in non-VC-backed firms. (See, for example, Hellmann and Puri [2002], who find that VCs play a significant role in the professionalization of start-up firms in general and in the hiring of their top managers in particular.) Similar arguments apply to institutional investors as well. We therefore eliminate VC-backed firms and firms that have shareholdings (greater than 5 percent) by

¹² We did not explicitly screen out firms with offer prices below \$5. However, we have only 29 such firms in our sample. Excluding these firms does not alter our results.

1993	1994	1995	1996	1997	1998	1999	2000	Total
615	516	539	792	473	307	499	376	4,117
91	108	85	119	24	12	2	3	444
98	68	40	64	36	26	39	30	401
48	29	31	62	71	67	44	10	362
10	4	17	25	3	1	34	37	131
40	12	10	12	3	0	0	5	82
0	0	2	8	7	9	2	1	29
3	4	1	4	3	2	1	1	19
0	2	0	1	0	1	0	1	5
128	106	169	253	155	100	288	239	1,438
72	60	82	108	51	38	45	31	487
125	123	102	136	119	52	44	18	719
	615 91 98 48 10 40 0 3 0 128 72	615 516 91 108 98 68 48 29 10 4 40 12 0 0 3 4 0 2 128 106 72 60	615 516 539 91 108 85 98 68 40 48 29 31 10 4 17 40 12 10 0 0 2 3 4 1 0 2 0 128 106 169 72 60 82	615 516 539 792 91 108 85 119 98 68 40 64 48 29 31 62 10 4 17 25 40 12 10 12 0 0 2 8 3 4 1 4 0 2 0 1 128 106 169 253 72 60 82 108	615 516 539 792 473 91 108 85 119 24 98 68 40 64 36 48 29 31 62 71 10 4 17 25 3 40 12 10 12 3 0 0 2 8 7 3 4 1 4 3 0 2 0 1 0 128 106 169 253 155 72 60 82 108 51	615 516 539 792 473 307 91 108 85 119 24 12 98 68 40 64 36 26 48 29 31 62 71 67 10 4 17 25 3 1 40 12 10 12 3 0 0 0 2 8 7 9 3 4 1 4 3 2 0 2 0 1 0 1 128 106 169 253 155 100 72 60 82 108 51 38	615 516 539 792 473 307 499 91 108 85 119 24 12 2 98 68 40 64 36 26 39 48 29 31 62 71 67 44 10 4 17 25 3 1 34 40 12 10 12 3 0 0 0 0 2 8 7 9 2 3 4 1 4 3 2 1 0 2 0 1 0 1 0 128 106 169 253 155 100 288 72 60 82 108 51 38 45	615 516 539 792 473 307 499 376 91 108 85 119 24 12 2 3 98 68 40 64 36 26 39 30 48 29 31 62 71 67 44 10 10 4 17 25 3 1 34 37 40 12 10 12 3 0 0 5 0 0 2 8 7 9 2 1 3 4 1 4 3 2 1 1 0 2 0 1 0 1 0 1 128 106 169 253 155 100 288 239 72 60 82 108 51 38 45 31

Table 1 Initial Public Offerings by Year

Note. CRSP = Center for Research in Security Prices.

financial institutions and corporations prior to the IPO (unless these entities are explicitly mentioned as wholly owned by firm insiders). This leaves us with 719 IPOs. (The information about shareholders is from the principal shareholders section of the IPO prospectuses.) Table 1 shows how we arrived at our final sample.

Various measures of management quality were hand collected from IPO prospectuses obtained from the Thomson Financial database. In particular, information on management team size and education level, former managerial experience, and tenure of the team members is from the management section of the prospectuses. Information on ATPs and internal governance mechanisms (such as chief executive officer [CEO]/chairman of the board duality, proportion of outside directors, and insider stock ownership) are obtained from the IPO prospectuses as well. Finally, stock returns are from CRSP, and accounting data are from Compustat.

4. Measures of Management Quality and Reputation and Firm Quality

4.1. Management Quality and Reputation

We use the following variables to measure the quality and reputation of a firm's management. First, management quality is affected by the amount of human and knowledge resources (including education and relevant work experience) available to a firm's management. This is measured by the number of executive officers and vice presidents on a firm's management team (TSIZE). Further, management quality depends on the education of management team members, which provides our second and third measures of management quality. We measure education in two ways, first, as the percentage of the management team with a master's of business administration (MBA) degree (PMBA) and, second, as the percentage of management team members who are certified public

accountants (PCPA). Higher percentages of MBAs and CPAs imply higher management quality.

Another contributing factor that increases management quality is relevant work experience, which provides our fourth and fifth measures of management quality. We measure work experience in two ways. First, we look at the percentage of management team members who have served as executive officers and/or vice presidents at other firms prior to joining the IPO firm (PFTEAM). Second, we look at the percentage of team members who have previously been partners in a law or accounting firm (PLAWACC). Clearly, expertise in law and accounting can be a useful asset to the firm at the time of going public and subsequently. In summary, the greater the value of these variables, the better the management quality.

The sixth measure of management quality we use is CEO dominance. On the one hand, a strong CEO may improve the cohesion of the management team. On the other hand, a strong-willed and dominating CEO may severely diminish possible contributions from other team members. Thus, while we believe that CEO dominance is an important measure of team quality, we are agnostic about the direction of the expected impact (positive or negative) of this measure. Our measure of CEO dominance is the ratio of CEO salary and bonus to the average salary and bonus of other team members listed in the executive compensation section of the prospectus in the fiscal year preceding the IPO (FCEO). Assuming that CEOs have a substantial influence over their own pay and nearly total influence over their subordinates' pay, this measure reflects the gap between the CEO's assessment of his own worth to the firm and his assessment of other team members' worth and is thus a good measure of CEO dominance.¹³

The seventh measure of management quality we use is the median tenure of the management team (TENURE), defined as the median number of years that team members have served with a firm. Longer median tenure may indicate cohesion and shared experiences and thus would imply lower transaction costs among team members.

However, longer tenures may also indicate the presence of complacency and rigidifying effects on team interactions. An ideal team would have members from different cohorts, which would allow for an inflow of new ideas and perspectives. Thus, higher management quality will be associated not only with a longer median tenure but also with a higher dispersion of tenures. Our eighth measure of management quality, therefore, is tenure heterogeneity, defined as the coefficient of variation of the team members' tenures (TENHET).

Table 2 summarizes our measures of management quality. The average (median) management team size (TSIZE) was 5.4 (5), with the smallest team consisting of only one person and the largest one of 15 members. On average, 7.5

¹³ Similar measures have been used in the strategy and organizational behavior literature to study the effect of management team quality on firm performance. D'Aveni (1990) and Hambrick and D'Aveni (1992) use such measures to study the deterioration of management team quality around bankruptcies.

	N	Mean	Median	Min	Max	SD
TSIZE	719	5.424	5.000	1.000	15.000	2.223
PMBA	719	.075	.000	.000	.800	.145
PCPA	719	.118	.077	.000	.833	.141
PFTEAM	719	.381	.333	.000	1.000	.277
PLAWACC	719	.031	.000	.000	.750	.088
FCEO	711	1.381	1.259	.000	5.425	.584
TENURE	719	6.482	5.000	1.000	30.500	5.239
TENHET	718	2.172	.940	.000	22.854	2.911
MQFACT	711	.000	099	-1.311	3.132	.703

Table 2 Summary Statistics of Management Quality Variables for Firms with Initial Public Offerings 1993–2000

percent of managers had an MBA degree (PMBA), 11.8 percent were CPAs (PCPA), 38.1 percent had held a top management position at another firm prior to joining the IPO firm (PFTEAM), and 3.1 percent have been a partner in a law or accounting firm (PLAWACC). On average, CEOs were earning 38.1 percent more than the average member of their management team (FCEO). The median tenure ranged from 1 to 30.5 years, with a mean (median) of 6.5 (5.0) years.

Although these variables are expected to measure management quality and reputation, they may have unique limitations as a measure of the underlying unobservable construct. Thus, we use common factor analysis to construct a single variable for management quality that captures variation common to the various observable proxies of management quality. In order to ensure that this common factor captures only the effect of management quality and not that of firm quality variables such as firm size, we use firm-size-adjusted variables to extract that factor. The management quality factor score (MQFACT) is obtained using common factor analysis on firm-size-adjusted TSIZE, MBA, CPA, FTEAM, LAWACC, and FCEO. These variables refer, respectively, to the number of management team members, the number of MBAs on the management team, the number of CPAs on the management team, the number of team members with prior managerial experience at other firms, the number of team members who served as partners in law and accounting firms, and CEO dominance. Table 2 reports summary statistics of MQFACT.

We exclude TENURE and TENHET from this common factor analysis since these variables, unlike the others, have negative factor loadings and negative scoring coefficients if included in the factor analysis. The interpretation of

¹⁴ A number of papers in the empirical finance and accounting literature make use of factor analysis to isolate the unobservable construct underlying several proxy variables. See, for example, Gaver and Gaver (1993) and Guay (1999), who make use of factor analysis to study the size of a firm's investment opportunity set.

¹is We adjust management quality variables for firm size by regressing each variable on firm size and then using the residuals from these regressions (that is, variation in management quality variables not explained by firm size) as firm-size-adjusted proxies of the management quality variables.

		b
Measure	Communalities	Correlation between MQFACT and Measure
TSIZE	.2320	.8382
MBA	.0394	.3379
CPA	.0756	.3856
FTEAM	.1941	.7281
LAWACC	.0666	.3377
FCEO	.0136	.0625

Table 3

Common Factor Analysis of Measures of Management Quality for Firms with Initial Public Offerings 1993–2000

MQFACT becomes problematic when some individual management quality variables have positive scoring coefficients and others have negative scoring coefficients. Thus, we restrict our common factor analysis to the first six individual management quality variables, since these have positive factor loadings and positive scoring coefficients, and use TENURE and TENHET as control variables in our multivariate analyses.¹⁶

Table 3 reports the results of this common factor analysis. It presents estimated starting communalities, calculated as the squared multiple correlations obtained from regressing each of the management quality measures on the other measures used in the factor analysis. As suggested by Harman (1976), the number of factors needed to approximate the original correlations among individual measures is equal to the number of summed eigenvalues needed to exceed the sum of communalities. The sum of communalities is .6213, which is less than the eigenvalue for the first factor in the factor analysis (.7980), which suggests that one factor parsimoniously explains the intercorrelations among the individual measures. Table 3 also reports the correlations between MQFACT and the original measures of management quality.

4.2. Proxies for Other Aspects of Firm Quality and Internal Governance and Control Variables

In order to separate the effects of management quality and reputation from those of other dimensions of firm quality and internal governance, we control for these other dimensions using the following proxies.¹⁷

¹⁶ The negative factor loadings and negative scoring coefficients for TENURE and TENHET are driven mostly by the negative correlations between TENURE and TENHET and other management quality variables such as the percentage of management team members with prior managerial experience at other firms and the percentage of management team members with MBA degrees (the correlation coefficients between TENURE [TENHET] and PFTEAM are −.52 [−.20], and the correlation coefficients between TENURE [TENHET] and PMBA are −.05 [−.03]). Indeed, managers who have longer tenures with a firm are more likely to be bred internally rather than invited from outside and, thus, are less likely to have prior managerial experience at other firms. Similarly, managers who have spent a greater number of years with a firm are more likely to acquire their managerial skills internally rather than externally at an educational institution.

¹⁷ Similar proxies are used by Field and Karpoff (2002), a study of takeover defenses of IPO firms.

First, a common firm quality variable used in many IPO studies is firm size. We use the natural logarithm of the book value of a firm's assets immediately prior to its IPO as a proxy for the firm size (LNBVA). The larger the firm, the higher the firm quality. Second, we control for the proportion of outside directors (directors who are listed in the management section of the prospectus and are not executive officers of the company, founders, former employees, or anyone who is engaged in any kind of business dealings with the firm) on the board of directors (ODIR). There are two ways in which outside directors can influence firm quality. They may provide additional knowledge (inputs and perspectives) to the firm's management. They also provide linkages to external parties, such as underwriters, financial institutions, and auditors. The greater the proportion of outside directors, the higher the firm quality.

Third, we control for insider stock ownership, defined as a proportion of voting power owned by executive officers and directors both before and after (depending on the particular analysis we conduct) the IPO (INSIDERB and INSIDERA, respectively). Sufficiently large insider stock ownership may serve as a substitute for ATPs. Fourth, we control for CEO/chairman of the board duality (BOSS). This dummy variable is equal to one if a firm's CEO is also a chairman of its board of directors and zero otherwise. Separation of the roles of a CEO and a chairman of the board creates greater management accountability and improves internal governance and firm quality.²⁰ Fifth, we control for underwriter reputation, since underwriters care for their reputation with potential buyers of IPO shares and may bargain with IPO firms to include optimal provisions in their corporate charters. Our underwriter reputation measure (REP) is the lead underwriter's share of the total proceeds raised by all IPOs in 1993–2000 (similar to Megginson and Weiss [1991]).

Sixth, we control for a firm's growth options measured as the ratio of the sum of capital expenditures and research and development expenses to the book value of assets prior to IPO (CERDA). We control for growth options since, according to our long-term value creation hypothesis, firms with higher quality managers that have greater growth options are more likely to use ATPs to shield themselves from unwanted takeover attempts and implement long-term value-creating projects (hypothesis 2). Seventh, we control for a firm's leverage (LEV-ERAGE), which is the ratio of long-term debt to the book value of assets prior to IPO.²¹ Eighth, we control for the existence of state-level ATPs (STATELAW), which is a dummy variable equal to one if the state in which an IPO firm is

¹⁸ This measure of firm quality has been widely used in the literature (see, for example, Ritter 1984; Michaely and Shaw 1994).

¹⁹ Some studies in the corporate control literature show that outside directors enhance firm value (see, for example, Cotter, Shivdasani, and Zenner 1997; Borokhovich, Parrino, and Trapani 1996).

²⁰ See, for example, Yermack (1997), who shows that firms that separate the roles of a chief executive officer (CEO) and a chairman of the board receive higher valuation, and Rechner and Dalton (1991), who show that such firms outperform those with combined roles of CEO and chairman.

²¹ Field and Karpoff (2002) argue that a firm's leverage, like its size, may have an effect on its vulnerability to takeovers and on the value of ATPs to it.

10	7 7 1111115 1111111	imuai i aone	Oneimgo 17	2000	
	Mean	Median	Min	Max	SD
LNBVA	16.726	16.843	12.064	23.810	1.543
ODIR	.313	.333	.000	1.000	.264
INSIDERB	.813	.900	.056	1.000	.219
INSIDERA	.564	.592	.010	.983	.185
BOSS	.730	1.000	.000	1.000	.444
REP	.016	.001	.000	.175	.037
CERDA	.201	.081	.000	5.985	.450
LEVERAGE	.239	.140	.000	6.378	.400
STATELAW	.921	1.000	.000	1.000	.270
DELAWARE	.452	.000	.000	1.000	.498

Table 4
Summary Statistics of Firm Quality and Other Control Variables
for Firms with Initial Public Offerings 1993–2000

Note. N = 719.

incorporated has at least one state ATP described in Appendix A and zero otherwise. We include this variable since state ATPs can serve as substitutes for firm-level ATPs (see, for example, Karpoff and Malatesta 1989; Field and Karpoff 2002). Finally, we include a dummy variable equal to one if an IPO firm is incorporated in the state of Delaware and zero otherwise (DELAWARE). Since 45 percent of IPO firms in our sample are incorporated in Delaware, we include this variable to control for the effect that Delaware corporate legislation may have on the likelihood of adopting ATPs. Table 4 summarizes our measures of firm quality, internal governance, and control variables.

5. Empirical Tests and Results

5.1. Relationship between Management Quality and the Prevalence of Antitakeover Provisions

The long-term value creation hypothesis predicts that higher quality managers will be more likely to have a greater number of ATPs in their firms' corporate charters prior to going public, insulating themselves from unwanted takeover bids and implementing long-term value creating projects (hypothesis 1). The managerial entrenchment hypothesis, however, predicts the opposite: lower quality managers will be more likely to adopt a greater number of ATPs in their firms' corporate charters prior to going public, insulating themselves from the market for corporate control and consuming private benefits of control. We first present the results of our univariate tests and then analyze these relationships using multivariate regressions.

5.1.1. Univariate Tests

Table 5 reports the frequencies of individual firm-level ATPs for our sample of IPO firms and the results of our univariate tests of the relationship between

the quality of a firm's management and the prevalence of ATPs. (Firm-level ATPs are described in Appendix A.)

First of all, it is worth noting that the frequencies of most of the firm-level ATPs in our sample of IPO firms in 1993-2000 increased sharply compared with the frequencies reported by Field and Karpoff (2002) for their sample of IPO firms in 1988-92. This finding provides further evidence that over time IPO firms increased the use of ATPs in their corporate charters as reported in previous studies (see, for example, Bebchuk 2003).²² For example, frequencies of the following ATPs almost doubled from the 1988-92 sample to the 1993-2000 sample: a restriction on the ability of common shareholders to call shareholder meetings (from 15.4 to 37.3 percent), an advance-notice requirement to call shareholder meetings (from 19.1 to 41.2 percent), unequal voting rights (from 5.4 to 10.3 percent), and a requirement to remove directors only for cause (from 12.8 to 24.5 percent). There is also a sizeable increase in the frequencies of such provisions as a staggered/classified board (from 36.2 to 43.5 percent), a blankcheck preferred stock (from 85.1 to 93.6 percent), a stakeholder clause (from 4.0 to 7.0 percent), a restriction on action by written consent (from 23.0 to 26.0 percent), and a supermajority required to approve mergers (from 9.2 to 13.8 percent). On the other hand, frequencies of some other provisions decreased, for example, a supermajority required to replace directors (from 27.3 to 14.3 percent), a fair-price provision (from 9.7 to 3.5 percent), and a poison pill (from 2.3 to 1.4 percent).

To study the relationship between management quality and ATPs, we split our sample by the median management quality factor score and compare the frequencies of ATPs in these subsamples. We refer to the firms with below median management quality factor scores as "low-management-quality firms" and those with above median management quality factor scores as "high-management-quality firms."

We find that management quality significantly affects the prevalence of ATPs in IPO firms. In particular, high-management-quality firms have significantly more ATPs regulating shareholder meetings. For example, the proportions of high-management-quality firms with a restriction on the ability of common shareholders to call shareholder meetings, with an advance-notice requirement, and with a restriction on action by written consent are greater by about 10 percentage points than those of low-management-quality firms with the same provisions. These differences are statistically significant at the 1 percent level. Further, the proportions of high-management-quality firms with a staggered board and a supermajority requirement to remove directors are significantly larger than those of low-management-quality firms with the same provisions. These differences are significant at the 5 and 10 percent levels, respectively. On

²² The median book value of assets prior to IPO reported by Field and Karpoff (2002) is \$27.7 million, while it is \$20.7 million for our sample. This indicates that these two samples are roughly comparable and the differences in reported frequencies of ATPs cannot be attributed to differences in IPO firm sizes.

Table 5 Summary Statistics of Individual Firm-Level Antitakeover Provisions (ATPs) for Firms with Initial Public Offerings (IPOs) 1993–2000

	Fu	Full Sample	le	Abc	Above Median MQFACT	an	Belo M	Below Median MQFACT	ut	Difference in	Difference in
Provision	Mean	Mean Median SD	SD	Mean	Mean Median SD	SD	Mean 1	Mean Median SD	SD	Means t-Statistic	Ĭ
1. Antigreenmail	.0153	0	.1228	9500.	0	.0750	.0253	0	.1572	-2.126^{*}	-2.121^{*}
2. Blank-check preferred stock	.9360	-	.2449	.9493	-	.2197	.9298	1	.2559	1.091	1.091
3. Staggered board	.4353	0	.4961	.4761	0	.5001	.4017	0	.4909	2.001^{*}	1.997*
4. Fair price	.0348	0	.1833	.0282	0	.1657	.0421	0	.2012	-1.010	-1.010
5. Poison pills	.0139	0	.1172	.0169	0	.1291	.0112	0	.1056	.641	.641
6. Stakeholder clause	.0695	0	.2545	9290.	0	.2514	.0730	0	.2606	283	283
Shareholder meeting restrictions:											
7. Meetings called only by directors or executives	.3727	0	.4839	.4338	0	.4963	.3174	0	.4661	3.223**	3.202**
8. Supermajority required to call special meetings	.0181	0	.1333	.0169	0	.1291	.0197	0	.1390	274	275
9. Advance-notice requirement	.4117	0	.4925	.4620	0	.4993	.3652	0	.4822	2.630**	2.619**
10. Restrictions on action by written consent	.2601	0	.4390	.3070	0	.4619	.2163	0	.4123	2.764**	2.751**

Supermajority vote requirements:											
11. Supermajority required to approve mergers	.1377	0	.3448	.1324	0	.3394	.1461	0	.3537	526	526
12. Supermajority required to replace directors	.1433	0	.3506	.1690	0	.3753	.1180	0	.3230	1.944^{+}	1.940^{+}
13. Supermajority required to amend charter and bylaws	.2976	0	.4575	.3042	0	.4607	.2949	0	.4567	.270	.270
14. Unequal voting rights	.1029	0	.3041	.0958	0	.2947	.1124	0	.3163	723	724
Miscellaneous ATPs:											
15. Directors can be removed only for cause	.2448	0	.4303	.2704	0	.4448	.2219	0	.4161	1.502	1.500
16. Merger must be approved by inside directors	.0028	0	.0527	9500.	0	.0750	0000	0	0000	1.418	1.417
17. Restrictions on transfer of common stock	.0362	0	.1868	.0310	0	.1735	.0421	0	.2012	791	791
18. Restrictions on votes each shareholder may cast	7600.	0	.0983	.0085	0	.0917	.0112	0	.1056	376	376
19. Prohibition of cumulative voting for election of directors	.7413	-	.4382	.7239	_	.4477	.7612	1	.4269	-1.137	-1.136
Total number of a firm's ATPs	1.2837	4	2.5905 4.5042	1.5042	4	2.5505	4.1096	3	2.6145	2.037*	2.318*

Note. Values are means (frequencies) and medians of individual firm-level ATPs and the total number of a firm's ATPs. The management quality factor score (MQEACT) is obtained using common factor analysis on firm-size-adjusted TSIZE, MBA, CPA, FTEAM, LAWACC, and FCEO. Significance levels are based on the parametric *t*-test for the difference in means between the samples of firms with above or below median values for MQFACT and the Wilcoxon rank-sum test for the difference in medians between the two samples.

† Significant at the .10 level.

* Significant at the .05 level. ** Significant at the .01 level.

the other hand, the proportion of high-management-quality firms with antigreenmail provisions is significantly smaller than that of low-management-quality firms.

Table 5 also reports the means and medians of the total number of firm-level ATPs (1–19 in Appendix A). Firms with high-quality management teams have a significantly greater total number of ATPs than do firms with low-quality management teams. The mean (median) total number of ATPs of firms with above median management quality factor scores is 4.51 (4), and it is 4.11 (3) for firms with below median management quality factor scores; these differences are significant at the 5 percent level.

Our univariate tests in Table 5 indicate that higher management quality is associated with more ATPs in IPO firms, which provides support for the long-term value creation hypothesis (hypothesis 1) and contradicts the managerial entrenchment hypothesis.

5.1.2. Multivariate Tests

To study the relationship between management quality and the prevalence of ATPs in IPO firms, we estimate the following Poisson maximum-likelihood specification:

$$\begin{split} \text{ATP}_i &= \beta_0 + \beta_1 T \text{SIZE}_i + \beta_2 \text{PMBA}_i + \beta_3 \text{PCPA}_i + \beta_4 \text{PFTEAM}_i + \beta_5 \text{PLAWACC}_i \\ &+ \beta_6 \text{FCEO}_i + \beta_7 \text{TENURE}_i + \beta_8 \text{TENHET}_i + \beta_9 \text{CERDA}_i + \beta_{10} \text{INSIDERB}_i \\ &+ \beta_{11} \text{ODIR}_i + \beta_{12} \text{BOSS}_i + \beta_{13} \text{LNBVA}_i + \beta_{14} \text{LEVERAGE}_i + \beta_{15} \text{REP}_i \\ &+ \beta_{16} \text{STATELAW}_i + \beta_{17} \text{DELAWARE}_i + \text{YEAR}_i + \text{IND}_i + \epsilon_i, \end{split}$$

where the dependent variable is the total number of ATPs that a firm has prior to its IPO.²³ The terms YEAR and IND are year and industry (two-digit SIC code) dummies.

Regression 1 in Table 6 presents the results of the estimation of equation (1). We find that TSIZE, PMBA, and PCPA have a positive and significant impact on the total number of ATPs in IPO firms. The coefficient estimates of these independent variables are significant at the 5 percent level (except for PMBA, which is significant at the 1 percent level). We also find that firms with more

²³ It can be argued that unequal voting rights are significant takeover deterrents by themselves, and insiders of firms with unequal voting rights may not need additional ATPs for takeover protection. For example, in our sample, firms with unequal voting rights have significantly lower frequencies of staggered boards, restrictions on action by written consent, and requirements for directors to be removed for cause. However, at the same time, firms with unequal voting rights have significantly higher frequencies of requirements for meetings to be called only by directors or executives, advance-notice requirements, and restrictions on transfer of common stock. There are no statistically significant differences in frequencies of other ATPs between firms with versus without unequal voting rights. (These results are available on request.) This indicates that firms with unequal voting rights do adopt other ATPs in their corporate charters. Nevertheless, we repeated our analysis, excluding firms with unequal voting rights from our sample, and the results were similar to those reported here.

Table 6

Relationship between Management Quality and the Prevalence of Antitakeover Provisions (ATPs) for Firms with Initial Public Offerings 1993-2000

			ATP	6							ATPDUM	1			
	(1)	(2)		(3)		(4)		(5)	()	(9)		(7)		(8)	
Constant	-1.278^{*} (-2.35)	-1.207*	(-2.24)					-9.642^{**} (-7.04)	(-7.04)	-9.682** ((-7.28)				
MQFACT		_* 200.	(2.29)			.052	(1.71)			.400**	(2.89)			.357*	(2.47)
TSIZE				*020*	(1.98)				(2.31)				(2.17)		
PMBA		_		$.246^{+}$	(1.91)				(2.97)				(2.40)		
PCPA		_		.176	(1.15)				$(2.76)^{**}$				(2.42)		
PFTEAM		_		108	(-1.27)				(10)				52)		
PLAWACC		_		.405	(1.68)				(16.)				(1.40)		
FCEO		_		.041	(1.21)				(2.25)				(1.53)		
TENURE				.002	(.46)		(.90)		(19)	.011	(.56)		(10		(.43)
TENHET				.004	(.57)		(1.01)		(.94)	.043	(1.25)		(1.15)		(1.57)
CERDA				.030	(.51)		(.58)		(.93)	.166	(.80)		(.37)		(.28)
INSIDERB	081 (88)	073	(80)	102 ((-1.07)) 960'-	-1.01)	749^{+}	(-1.75)	710^{+}	(-1.70)	707 (-	-1.58)	694 (-	-1.57)
ODIR				.071	(.91)		(.87)		(53)	151	(45)		(20.		(03)
BOSS				.003	(90.)		(.11)		(47)	097	(49)		(47)		(52)
LNBVA				.092**	(5.00)	*	(6.52)		(5.02)	.482**	(6.56)	*	(4.58)		(5.79)
LEVERAGE				1111+	(-1.81)	$\overline{}$	-2.00		(35)	.010	(.05)		(19)		(40)
REP	.282			.532	(66.)		(1.15)		(1.63)	4.636	(1.54)		(1.65)		(1.59)
STATELAW	.563**								(4.92)	1.961**	(4.93)				
DELAWARE	.040								(19.)	.044	(.25)				
State fixed effects		No		Yes		Yes				No		Yes		Yes	
Industry dummies	Yes	Yes		Yes		Yes		No		No		No		No	
N	711	711		707	7	707		711		711	~	829	9	78	
Pseudo R²	.0925	.0871						.1855		.1644		.1545		.1346	
			•	::											

Note. Columns 1–4 present the results of Poisson maximum likelihood regressions; columns 5–8 present the results of logit regressions. Dependent variables are ATP, which is the total number of firm-level ATPs (1–19 in Appendix A), and ATPDUM, which is equal to one if a firm has four or more ATPs and zero otherwise. All regressions include year dummies. Values in parentheses are *t*-statistics.

⁺ Significant at the .10 level.

* Significant at the .05 level.

** Significant at the .01 level.

dominating CEOs (FCEO) have a greater number of ATPs. The coefficient estimate of this variable is significant at the 10 percent level. These results are also economically significant. For a firm with the median number of four ATPs, increases of 1 standard deviation in management team size (TSIZE) and the percentage of MBAs in a management team (PMBA) lead to an increase in the number of ATPs by .21; this represents a 5.2 percent increase over the median number of ATPs. In addition, for a firm with the median number of four ATPs, an increase of 1 standard deviation in the percentage of CPAs in a management team (PCPA) and CEO dominance (FCEO) lead to increases in the number of ATPs by .18 and .14, respectively; these represent increases of 4.2 and 3.5 percent over the median number of ATPs, respectively.

In regression 2, we replace individual management quality variables with MQFACT. This regression also shows a significant positive relationship between management quality and the number of ATPs; the coefficient estimate of MQFACT is positive and significant at the 5 percent level. For a firm with the median number of four ATPs, an increase of 1 standard deviation in MQFACT leads to an increase in the number of ATPs by .19; this represents a 4.9 percent increase over the median number of ATPs.

In regressions 3 and 4, we drop STATELAW and DELAWARE and estimate state-of-incorporation fixed-effects models. The results are qualitatively similar to those of regressions 1 and 2; TSIZE, PMBA, PLAWACC, and MQFACT have a positive and significant effect on the total number of ATPs.

Next we repeat the analysis using logit regressions with a dependent variable equal to one if an IPO firm has four or more ATPs (firms with stronger antitakeover protection) and zero otherwise (firms with weaker antitakeover protection).²⁴ The results of these logit regressions (regressions 5–8) confirm the results of our Poisson maximum-likelihood estimations. In regression 5, with individual management quality variables, PMBA, PCPA, TSIZE, and FCEO have a positive and significant impact on the prevalence of ATPs in IPO firms. The coefficient estimates of PMBA and PCPA are significant at the 1 percent level, and the coefficient estimates of TSIZE and FCEO are significant at the 5 percent level. These results are also economically significant. For a firm facing equal probabilities of having stronger versus weaker antitakeover protection, increases of 1 standard deviation in PMBA, PCPA, TSIZE, and FECO lead to increases of 6.9, 6.3, 6.0, and 5.3 percentage points in the probability of having stronger antitakeover protection, respectively.

In regression 6, we use MQFACT in place of individual management quality variables; MQFACT has a positive and significant influence on the prevalence of ATPs, with its coefficient estimate significant at the 1 percent level. For a firm facing equal probabilities of having stronger versus weaker antitakeover protec-

²⁴ We use four or more ATPs as a cutoff number since it roughly divides the sample into two equal parts; there are 338 firms in our sample with three or fewer ATPs in their corporate charters and 381 firms with four or more ATPs in their corporate charters. Our results are generally robust to other cutoff numbers.

tion, an increase of 1 standard deviation in the management quality factor score leads to an increase of 7.0 percentage points in the probability of having stronger antitakeover protection.

In regressions 7 and 8, we drop STATELAW and DELAWARE and estimate state-of-incorporation fixed-effects models. The results are qualitatively similar to those of regressions 5 and 6; TSIZE, PMBA, PCPA, and MQFACT have a positive and significant effect on the prevalence of ATPs in IPO firms.²⁵

Both our univariate and multivariate tests of the relationship between management quality and the prevalence of ATPs in IPO firms show that firms with higher quality managers have a significantly greater number of ATPs prior to IPO. These findings support the long-term value creation hypothesis (hypothesis 1) and contradict the managerial entrenchment hypothesis.

5.2. Relationship among Management Quality, Growth Options, and the Prevalence of Antitakeover Provisions

The long-term value creation hypothesis predicts that stronger ATPs are more likely to be adopted by higher quality managers of firms that have larger growth options (firms with the greatest potential for long-term value creation) and are less likely to be adopted by lower quality managers of firms that have smaller growth options (firms with the smallest potential for long-term value creation; hypothesis 2).

To study the interaction between management quality and growth options and their impact on the prevalence of ATPs, we split our sample into four groups. In group 1 we place firms with above median management quality factor score (MQFACT) and above median growth options (CERDA). In group 2 we place firms with above median management quality factor score and below median growth options. In group 3 we place firms with below median management quality factor score and above median growth options. Finally, in group 4 we place firms with below median management quality factor score and below median growth options. We create dummies for each group and use them as independent variables (instead of management quality and growth option variables) in equation (1).

Table 7 reports the results of Poisson maximum-likelihood regressions of the total number of ATPs on the management quality and growth option group dummies and control variables. In regression 1, we include only the dummy variable for group 1. The coefficient estimate is positive and statistically significant at the 5 percent level. This indicates that firms with higher management quality

²⁵ We also investigated how management quality affects the likelihood of having individual ATPs (these results are available on request). For example, MQFACT has a significant and positive influence on the likelihood of IPO firms having a staggered board, a restriction on shareholders' ability to call shareholder meetings, an advance-notice requirement, and a supermajority requirement to remove directors. These results confirm the positive relationship between management quality and the prevalence of ATPs in IPO firms' corporate charters on both the aggregate and the individual provision levels.

	(1)	(2)	(3)	(4)
Constant	-1.371* (-2.55)	-1.240* (-2.32)		
GROUP1	.111* (2.44)		.100* (2.13)	
GROUP2		066 (-1.17)		068 (-1.19)
GROUP3		075 (-1.34)		050 (85)
GROUP4		219** (-3.75)		207** (-3.45)
TENURE	.003 (.67)	.005 (1.11)	.002 (.60)	.004 (.86)
TENHET	.004 (.58)	.005 (.67)	.007 (1.01)	.008 (1.11)
INSIDERB	067 (73)	071 (77)	087 (92)	090 (95)
ODIR	.007 (.10)	011 (15)	.058 (.74)	.038 (.49)
BOSS	.007 (.16)	.002 (.06)	.012 (.26)	.005 (.11)
LNBVA	.130** (8.21)	.129** (8.08)	.114** (6.90)	.113** (6.76)
LEVERAGE	063 (-1.16)	067 (-1.25)	114^{+} (-1.93)	117* (-1.99)
REP	.379 (.72)	.270 (.52)	.629 (1.18)	.530 (.99)
STATELAW	.556** (5.91)	.564** (5.97)		
DELAWARE	.040 (1.02)	.035 (.88)		
State fixed effects	No	No	Yes	Yes
N	711	711	707	707
Pseudo R ²	.0871	.0898		

Table 7 Relationship among Management Quality, Growth Options, and the Prevalence of Antitakeover Provisions (ATPs) for Firms with Initial Public Offerings 1993-2000

Note. Coefficients are from Poisson maximum-likelihood regressions with the dependent variable equal to the total number of ATPs (1-19 in Appendix A). All regressions include year and industry dummies. GROUP1 through GROUP4 are dummy variables representing four management quality and growth option groups. Values in parentheses are t-statistics.

** Statistically significant at the .01 level.

and larger growth options have a significantly greater number of ATPs than the rest of the sample. This result is also economically significant. A firm with the median number of four ATPs in groups 2, 3, and 4 will have .47 more ATPs if moved to group 1; this represents an 11.7 percent increase over the median number of ATPs.

In regression 2, we use group 2, 3, and 4 dummies. The coefficient estimates of these dummies are -.07, -.08, and -.22, respectively, and only the coefficient estimate of the group 4 dummy is significant at the 1 percent level. This indicates that firms in those groups have fewer ATPs than firms in group 1 and that firms in group 4 have significantly fewer ATPs than firms in group 1. This result is also economically significant. A firm with the median number of four ATPs in group 1 will have .79 fewer ATPs if moved to group 4; this represents a 19.6 percent decrease over the median number of ATPs.

In regressions 3 and 4, we drop STATELAW and DELAWARE and estimate state-of-incorporation fixed-effects models. The results are qualitatively similar to those of regressions 1 and 2; the coefficient estimate of the group 1 dummy is positive and significant in regression 3, and that of the group 4 dummy is negative and significant in regression 4. Thus, our findings in Table 7 provide support for hypothesis 2.

Statistically significant at the .10 level. * Statistically significant at the .05 level.

5.3. Relationship among Management Quality, Antitakeover Provisions, and Post–Initial Public Offering Operating Performance

The long-term value creation hypothesis predicts that firms with higher quality managers that have a greater number of ATPs will use this protection to implement long-term value-creating projects. Implementation of these projects will be reflected in a better post-IPO operating performance for such firms compared with the rest of the firms in the IPO sample (hypothesis 3). The managerial entrenchment hypothesis, on the other hand, predicts that regardless of management quality, firms associated with a greater number of ATPs will be more likely to have poorer post-IPO operating performance because of the loss-of-discipline effect.

We use four measures of operating performance: OIBD/Assets, ROA, profit margin, and OIBD/Sales, where OIBD is the operating income before depreciation plus interest income (Compustat items 13 and 15), Assets are the book value of total assets (item 6), ROA is the ratio of net income (item 172) over assets, and profit margin is the ratio of net income over sales (item 12). We further adjust these measures for industry performance by subtracting contemporaneous industry (two-digit SIC code) medians.

We split our sample into two roughly equal groups: firms with stronger antitakeover protection and firms with weaker antitakeover protection. Table 8 compares industry-adjusted median operating performance measures of firms in these two groups for the year of IPO (year 0) and the subsequent 4 years.

Consistent with the findings in the previous literature, the operating performance of all IPOs deteriorates in the years after the IPO, irrespective of the level of antitakeover protection.²⁶ However, firms with stronger antitakeover protection consistently and significantly outperform firms with weaker antitakeover protection (median tests are significant at the 1 percent level) in all years after the IPO. These results contradict the managerial entrenchment hypothesis.

To test the hypothesis that higher quality managers use ATPs to ward off unwanted takeovers and implement long-term value-creating projects (hypothesis 3), we split our sample into two groups. In group A we place firms with an above median management quality factor score and four or more ATPs. We place the rest of the sample in group B. In other words, group A represents firms with higher quality managers and stronger antitakeover protection. We then compare the post-IPO operating performances of group A and group B.

Table 9 compares industry-adjusted median operating performance measures of firms in group A with those of firms in group B for the year of IPO (year 0) and 4 years afterward. The median industry-adjusted operating performance measures of firms in group A are consistently greater than those of firms in

²⁶ See, for example, Jain and Kini (1994) and Mikkelson, Partch, and Shah (1997), who document declining operating performance for IPO firms in the years after going public and provide various explanations such as insiders timing their issues to follow periods of good performance, increased agency costs, window dressing accounting numbers prior to going public, and increased asset base.

Table 8
Relationship between Antitakeover Provisions (ATPs) and Post–Initial Public Offering (IPO) Operating Performance for Firms with IPOs 1993–2000

	U	ntitakeover ection	Weaker An Prote		Difference in Medians
Performance Measure	%	N	%	N	z-Statistic
OIBD/Assets 0	5.42	371	2.61	332	4.143**
OIBD/Assets 1	3.40	354	-1.05	313	4.610**
OIBD/Assets 2	3.59	319	-1.18	282	4.289**
OIBD/Assets 3	3.54	294	76	262	4.409**
OIBD/Assets 4	3.39	259	.85	225	3.090**
ROA 0	4.56	371	2.78	332	3.278**
ROA 1	3.10	355	.33	314	4.484**
ROA 2	2.09	320	85	283	4.316**
ROA 3	2.66	295	-1.03	265	5.153**
ROA 4	1.73	261	.03	227	3.344**
Profit margin 0	2.88	369	1.98	325	3.271**
Profit margin 1	2.27	355	02	311	4.035**
Profit margin 2	1.44	319	89	282	4.057**
Profit margin 3	2.08	294	-1.00	261	5.333**
Profit margin 4	1.30	261	24	225	3.384**
OIBD/Sales 0	4.25	369	1.37	325	4.272**
OIBD/Sales 1	3.04	354	56	310	4.120**
OIBD/Sales 2	2.65	318	-1.40	281	4.679**
OIBD/Sales 3	3.54	293	23	259	4.584**
OIBD/Sales 4	2.21	259	.00	222	3.706**

Note. Values are median levels of industry-adjusted operating performance of IPO firms by the prevalence of ATPs. Firms with stronger antitakeover protection have four or more ATPs; firms with weaker antitakeover protection have three or fewer ATPs. All performance measures are adjusted for industry performance by subtracting contemporaneous industry (two-digit Security Industry Classification code) medians. Year 0 is the year of IPO. Significance levels are based on the Wilcoxon rank-sum test for the difference in medians. ** Significant at the .01 level.

group B in the year of IPO and all of the following 4 years across all four measures of operating performance. Firms in group A significantly outperform firms in group B in at least 3 out of 5 years after the IPO (including the year of IPO). For example, firms in group A significantly outperform firms in group B in years 0, 3, and 4 based on ROA, profit margin, and OIBD/Sales measures and in years 1, 3, and 4 based on the OIBD/Assets measure.

These results provide support for the long-term value creation hypothesis (hypothesis 3). Along with our results from Section 5.2, where we presented evidence that higher management quality firms use a greater number of ATPs in their corporate charters when they have long-term value-increasing projects (larger growth options), these results suggest that IPO firms with higher quality managers use ATPs to protect themselves against potential takeovers and implement long-term value-creating projects, thus obtaining better post-IPO operating performance than the rest of the IPO firms.

Table 9

Relationship among Management Quality, Antitakeover Provisions (ATPs), and Post–Initial Public Offering (IPO) Operating Performance for Firms with IPOs 1993–2000

	Grou	ар А	Grou	ир В	Difference in
Performance Measure	%	N	%	N	Medians z-Statistic
OIBD/Assets 0	4.55	202	3.98	494	1.413
OIBD/Assets 1	3.40	190	1.38	469	1.955+
OIBD/Assets 2	1.85	168	.92	426	1.439
OIBD/Assets 3	3.50	155	1.35	395	2.205*
OIBD/Assets 4	3.46	133	1.64	345	1.650^{+}
ROA 0	4.21	202	3.27	494	1.752+
ROA 1	2.87	191	1.49	470	1.353
ROA 2	1.77	168	.63	428	1.092
ROA 3	2.88	155	.57	399	2.555*
ROA 4	1.88	134	.65	348	1.972*
Profit margin 0	3.09	201	2.31	487	2.215*
Profit margin 1	1.78	191	1.14	467	.566
Profit margin 2	1.18	167	.31	427	.615
Profit margin 3	2.31	154	.19	396	2.849**
Profit margin 4	1.92	134	.28	346	1.783+
OIBD/Sales 0	4.91	201	2.51	487	2.653**
OIBD/Sales 1	2.51	190	1.03	466	1.514
OIBD/Sales 2	2.20	167	.24	425	1.465
OIBD/Sales 3	3.32	154	.73	393	2.417*
OIBD/Sales 4	2.03	133	.95	342	1.719^{+}

Note. Values are median levels of industry-adjusted operating performance of IPO firms. Group A represents firms with above median management quality factor scores and four or more ATPs. Group B represents the rest of the sample. All performance measures are adjusted for industry performance by subtracting contemporaneous industry (two-digit Security Industry Classification code) medians. Year 0 is the year of IPO. Significance levels are based on the Wilcoxon rank-sum test for the difference in medians.

5.4. Relationship among Management Quality, Antitakeover Provisions, and Initial Public Offering Firm Valuation

According to the long-term value creation hypothesis, IPO firms with a greater number of ATPs and higher management quality will receive a higher valuation at the IPO stage than will the rest of the sample (hypothesis 4), since the market perceives that such firms will use antitakeover protection to implement long-term value-creating projects. In contrast, the managerial entrenchment hypothesis predicts that, regardless of management quality, IPO firms with a greater number of ATPs will receive a lower IPO valuation since ATPs tend to entrench management at the expense of firm shareholders.

We measure IPO firm valuation by Tobin's *q*, which is the ratio of the market value of assets to the book value of assets, where the market value of assets is equal to the book value of assets minus the book value of common equity plus the number of shares outstanding multiplied by the IPO offer price. We also use industry-adjusted Tobin's *q*, which is the difference between the IPO firm's

Significant at the .10 level.

^{*} Significant at the .05 level.

^{**} Significant at the .01 level.

Tobin's q and the median of its two-digit SIC code industry peers. We measure Tobin's q for industry peers the same way as for IPO firms and calculate their market value of assets using the share price at the end of the issue month. The book value of assets, the book value of equity, and the number of shares outstanding for both IPO firms and industry peers are taken from the first available post-IPO quarter on Compustat.

We run ordinary least squares (OLS) regressions of IPO firms' Tobin's *q* and industry-adjusted Tobin's *q* on ATP or ATPDUM and other control variables, where ATP is the total number of ATPs (1–19 in Appendix A) and ATPDUM is a dummy variable equal to one if a firm has four or more ATPs and zero otherwise. Table 10 reports the results of these regressions. The coefficient estimates of ATP and ATPDUM are positive in all specifications, although not statistically significant. These findings indicate that, although there is a positive relationship between the strength of antitakeover protection and IPO firm valuation, ATPs alone do not significantly influence IPO firm valuation and thus do not provide support for the managerial entrenchment hypothesis.

We test the long-term value creation hypothesis (hypothesis 4) by running OLS regressions of IPO firms' Tobin's q and industry-adjusted Tobin's q on GROUPADUM and control variables, where GROUPADUM is a dummy variable equal to one for firms with above median management quality factor scores and four or more ATPs (in other words, GROUPADUM is equal to one for firms in group A as described in Section 5.3) and zero for the rest of the sample. Table 11 reports the results of these regressions. In regression 1, where we use Tobin's q as a dependent variable, GROUPADUM has a positive coefficient estimate of .22, which is significant at the 5 percent level. This indicates that IPO valuations of firms in group A are significantly larger (by .22) than those of firms in group B. This difference represents a 9 percent increase over the median Tobin's q.

In regression 2, where we use industry-adjusted Tobin's q as a dependent variable, GROUPADUM has a positive coefficient estimate of .20, which is significant at the 5 percent level. This indicates that IPO valuations of firms in group A are significantly larger than those of firms in group B by .20. This difference represents a 28 percent increase over the median industry-adjusted Tobin's q.

In regressions 3 and 4, we drop STATELAW and DELAWARE and estimate state-of-incorporation fixed-effects models. The results are qualitatively similar to those of regressions 1 and 2; the coefficient estimates of GROUPADUM are positive and significant at the 10 percent level.

These results provide support for the long-term value creation hypothesis and indicate that firms with higher management quality and a greater number of ATPs receive a significantly higher valuation at the time of IPO than does the rest of the sample.

Table 10

Relationship between Antitakeover Provisions (ATPs) and Initial Public Offering (IPO) Firm Valuation for Firms with IPOs 1993-2000

justed 9	(6.42)	(.42)	(2.18)	-1.51)	1.40)	(2.31)	1.15)	59)	(.17)	-5.41)	(.74)	(3.87)							equity equity value ter on lustry
Industry-Adjusted Tobin's <i>q</i> (8)			_	$\overline{}$	$\overline{}$					*	∞				Yes	.0	0	1087	d as the nmon ce book O quart
Indus T	3.987**	.043	.151	015	022	.300	.245	105	.017	*66I	880.	5.182**			Ye	No	Z	.10	easurec of con rrs). Th oost-IP(
l's <i>q</i>	(3.21)	(60.)	(2.90)	(-1.15)	(86)	(3.83)	(.22)	(67)	(.42)	(-4.79)	(1.40)	(3.41)							s q is m ok value stry pee ailable p
Tobin's <i>q</i> (7)	4.236**	600.	.214**	012	017	.483**	.049	122	.044	193**	.171	4.808**			Yes	Yes	Yes	.3257	es. Tobin's us the boo I (for indu ne first ave Industry (
djusted s q	(6.53) (1.36)		(2.13)	-1.54)	-1.39)	(2.33)	(1.19)	(63)	(.14)	-5.65	(.81)	(3.79)							t variable ssets plus e month from the Security $N = 70$
Industry-Adjusted Tobin's <i>q</i> (6)	4.040**		.147*	015	022	.303*	.255	112	.015	207** (960.	5.071**			Yes	No	No	.1109	from ordinary least squares regressions with Tobin's q and industry-adjusted Tobin's q as dependent variables. Tobin's q is measured as the ratio f assets to the book value of assets, where the market value of assets is equal to the book value of assets plus the book value of common equity shares outstanding times the offer price (for IPO firms) or times the share price at the end of the issue month (for industry peers). The book value ue of equity, and the number of shares outstanding both for IPO firms and industry peers are taken from the first available post-IPO quarter on edjusted Tobin's q is the difference between the IPO firm's Tobin's q and the median of its two-digit Security Industry Classification code industry adustry-adjusted Tobin's q are winsorized at the 99th percentile. Values in parentheses are t -statistics. $N = 704$.
	(3.26) (1.01)		(2.84)	-1.18)	-1.01)	(3.81)	(.26)	(70)	(.41)	-5.01)	(1.48)	(3.36)							in's q as he book the end stry peer an of its eses are
Tobin's <i>q</i> (5)	4.297** .020		.209**	012	018	.479**	.058	128	.044	202** (.181	4.738**			Yes	Yes	Yes	.3268	iusted Tob equal to t re price at and indus I the medi
sdjusted 's q	(888)	(.67)	(2.46)	-1.25)	-1.60)	(2.11)	(1.22)	(45)	(14)	-5.58)	(.55)	(3.83)	-1.95)	(1.29)					ustry-adassets is a sthe share of firms is a and in's a and in's a and in's a stress is a
Industry-Adjusted Tobin's <i>q</i> (4)	4.179**	.063	.164*	012	025	.244*	.248	076	014	195** (.063	5.005**	329^{+}	.119	No	No	No	.1136	q and inc t value of is) or time both for Il firm's Tobi percentile
b s,	(3.10)	(39)	(3.21)	-1.08)	-1.21)	(3.32)	(09.)	(51)	(.10)	-4.76	(68.)	(3.42)	(05)	(1.14)					Tobin's re marke IPO firm tanding he IPO the 19th the 19t
Tobin's <i>q</i>	4.023**	.038	.229**	011 (021	.376**	.128	088	.010	183** (.105	4.701**	009	.107	No	Yes	Yes	.3255	sions with , where th price (for J nares outst between th
Adjusted 's q	(7.07) (1.56)		(2.40)	(-1.29)	(-1.58)	(2.13)	(1.25)	(48)	(16)	-5.83)	(65.)	(3.77)	(-2.14)	(1.26)					res regres of assets he offer j hber of sh ifference
Industry-Adjusted Tobin's <i>q</i> (2)	4.265** .029		.159*	012	024	.245*	.255	081	016	203** (.067	4.920**	363*	.116	No	No	No	.1161	least squarook value ing times the nun q is the d look of I Tobin's q
<i>b</i> s	(3.18) (1.28)		(3.15)	-1.12)	-1.21)	(3.29)	(.63)	(52)	(60.)	-5.00	(.95)	(3.39)	(23)	(1.11)					rdinary to the b sutstandi quity, an Tobin's
Tobin's q	4.129** .024		.225**	011 (-	021	.372**	.133	060.—	600.	192** (·	.112	4.652**	040	.103	No	Yes	Yes	.3271	are from o of assets of shares o value of ec y-adjusted
	Constant ATP	ATPDUM	MQFACT	TENURE	TENHET	CERDA	INSIDERB	ODIR	BOSS	LNBVA	LEVERAGE	REP	STATELAW	DELAWARE	State fixed effects	Year dummies	Industry dummies	Pseudo R ²	Note. Coefficients are from ordinary least squares regressions with Tobin's <i>q</i> and industry-adjusted Tobin's <i>q</i> as dependent variables. Tobin's <i>q</i> is measured as the ratio of the market value of assets is equal to the book value of assets but the book value of common equity minus the number of shares outstanding times the offer price (for IPO firms) or times the share price at the end of the issue month (for industry peers). The book value of assets, the book value of equity, and the number of shares outstanding both for IPO firms and industry peers are taken from the first available post-IPO quarter on Compustat. Industry-adjusted Tobin's <i>q</i> is the difference between the IPO firm's Tobin's <i>q</i> and the median of its two-digit Security Industry Classification code industry peers. Tobin's <i>q</i> and industry-adjusted Tobin's <i>q</i> are winsorized at the 99th percentile. Values in parentheses are <i>t</i> -statistics. <i>N</i> = 704.

* Significant at the .05 level.

	Tobin's q	Industry-Adjusted Tobin's q	Tobin's q	Industry-Adjusted Tobin's q
	(1)	(2)	(3)	(4)
Constant	3.802** (2.93)	4.150** (6.95)	4.032** (3.05)	3.990** (6.46)
GROUPADUM	.221* (2.15)	.202* (2.05)	.182+ (1.69)	.173+ (1.68)
TENURE	018^{+} (-1.88)	016^{+} (-1.80)	019^{+} (-1.92)	020* (-2.05)
TENHET	021 (-1.24)	026^{+} (-1.72)	018 (-1.00)	024 (-1.49)
CERDA	.374** (3.29)	.251* (2.18)	.487** (3.85)	.313* (2.41)
INSIDERB	.131 (.61)	.239 (1.17)	.045 (.20)	.227 (1.06)
ODIR	115 (66)	103 (61)	154 (84)	133 (75)
BOSS	.025 (.25)	005 (05)	.059 (.55)	.027 (.26)
LNBVA	177**(-4.73)	191** (-5.64)	189^{**} (-4.79)	198** (-5.52)
LEVERAGE	.107 (.91)	.061 (.53)	.172 (1.41)	.087 (.73)
REP	4.830** (3.50)	5.086** (3.89)	4.925** (3.48)	5.259** (3.93)
STATELAW	046 (27)	347* (-2.09)		
DELAWARE	.115 (1.23)	.127 (1.38)		
State fixed effects	No	No	Yes	Yes
Year dummies	Yes	No	Yes	No
Industry dummies	Yes	No	Yes	No
Pseudo R ²	.3189	.1100	.3194	.1055

Table 11

Relationship among Management Quality, Antitakeover Provisions (ATPs), and Initial Public Offering (IPO) Firm Valuation for Firms with IPOs 1993–2000

Note. Coefficients are the results of ordinary least squares regressions with Tobin's q and industry-adjusted Tobin's q as dependent variables. Tobin's q and industry-adjusted Tobin's q are winsorized at the 99th percentile. Values in parentheses are t-statistics. N=704.

5.5. Relationship among Management Quality, Antitakeover Provisions, and Post–Initial Public Offering Stock Return Performance

The long-term value creation hypothesis predicts that firms with higher quality managers and stronger antitakeover protection will use such defenses to implement long-term value-creating projects. This will be reflected in a better post-IPO stock return performance for such firms (hypothesis 5). In contrast, the managerial entrenchment hypothesis predicts that regardless of management quality, firms with stronger antitakeover protection will have poorer post-IPO long-term stock returns because of the loss-of-discipline effect.

We study stock return performance of IPO firms relative to various benchmarks: NASDAQ equal- and value-weighted indices, CRSP equal- and value-weighted indices, and the Standard & Poor's 500 Index. We calculate a 5-year holding period return for each IPO firm and corresponding benchmarks. We define a year as 12 intervals of 21 trading days (252 days). Holding period returns for each firm and a corresponding benchmark are calculated as $[\prod_{t=1Ti}(1+R_{ii})-1] \times 100$ percent, where R_{ii} is the return on IPO firm stock i or a corresponding benchmark on the t-th day of the 5-year event window and T_i is the number of trading days in the event window. The term T_i is equal to 1,260 if

⁺ Significant at the .10 level.

^{*} Significant at the .05 level.

^{**} Significant at the .01 level.

an IPO firm survives for 5 years after the issue. For firms that were delisted before the end of the 5-year window, the holding period return is calculated until the delisting date, and the corresponding benchmark's return is calculated over the same truncated period. The first day after the issue date is t=1. We do not include the issue day return (that is, initial return or underpricing), as it is frequently difficult for an ordinary investor to buy a share of an IPO firm at the offering price.²⁷ Finally, we calculate the abnormal holding period return for each firm by subtracting the 5-year holding period return of the corresponding benchmark from the 5-year holding period return of the IPO firm.

We split our sample into two groups as in the case of operating performance (firms with stronger antitakeover protection, or those with four or more ATPs, and firms with weaker antitakeover protection, or those with three or less ATPs) and test the differences in abnormal holding period returns between the two groups.

In Table 12 we report the mean and median 5-year holding period returns for IPO firms and corresponding benchmarks split into two groups by the prevalence of ATPs. Consistent with the findings in the previous literature, IPO firms underperform corresponding benchmarks in 5 years after going public since both the mean and median holding period returns of IPO firms are less than those of corresponding benchmarks (see, for example, Ritter 1991; Loughran and Ritter 1995; Brav and Gompers 1997; Teoh, Welch, and Wong 1998). However, both the mean and median abnormal holding period returns of IPO firms with stronger antitakeover protection are significantly larger than those of IPO firms with weaker antitakeover protection. The differences in mean abnormal holding period returns are statistically significant at the 10 percent level, and the differences in median abnormal holding period returns are significant at the 1 and 5 percent levels. These results contradict the managerial entrenchment hypothesis.

To test the long-term value creation hypothesis (hypothesis 5), we divide our sample into two groups, A and B, as described in Section 5.3. In group A we place firms with above median management quality factor score and with four or more ATPs. We place the rest of the sample in group B. Thus, firms in group A have higher quality managers and stronger ATPs. In Table 13, we report the mean and median 5-year holding period returns for IPO firms and corresponding benchmarks. We find that IPO firms in group A have consistently larger abnormal holding period returns than IPO firms in group B across all benchmarks. The differences in mean abnormal holding period returns are significant at the 1, 5, and 10 percent levels depending on the benchmark used, and the differences in median abnormal holding period returns are all significant at the 1 percent level. These results indicate that firms in group A have better post-IPO stock return

²⁷ Including the issue day return in our analysis does not change the results.

Relationship between Antitakeover Provisions (ATPs) and Post-Initial Public Offering (IPO) Stock Return Performance for Firms with IPOs 1993-2000 Table 12

	Stı	ronger Antita $(N =$	Stronger Antitakeover Protection $(N = 381)$	ion	×	eaker Antitak (N =	Weaker Antitakeover Protection $(N = 338)$	ion		
	Mean IDO	Mean	Median Median Median Mean Median Median Median Median Median Median IDO Renchmark Mean IDO Renchmark Median IDO Renchmark	Median	Mean IPO	Mean	Median IDO	Median	Difference in Mean Abnormal Holding Period Returns	Difference in Median Abnormal Holding Period Returns
Benchmark	Return	Return	Return	Return	Return	Return	Return	Return		z-Statistic
NASDAQ EW Index	26.94	312.56	-40.60	296.78	18.30	336.47	-55.64	309.71	32.56+ (1.777)	34.28* (2.324)
NASDAQ VW Index	26.94	103.01	-40.60	75.10	18.30	126.55	-55.64	111.47	32.19^{+} (1.960)	36.05^{**} (3.645)
CRSP EW Index	26.94	191.26	-40.60	191.59	18.30	204.30	-55.64	200.09	21.68 (1.352)	19.90^{*} (2.562)
CRSP VW Index	26.94	80.22	-40.60	71.92	18.30	60.86	-55.64	100.01	26.52^{+} (1.685)	28.46^{**} (3.809)
S&P 500 Index	26.94	76.93	-40.60	96.69	18.30	95.71	-55.64	99.75	27.43^{+} (1.738)	32.72** (3.904)

Note. Data are percentages unless otherwise specified. Returns are calculated by compounding daily returns for 1,260 trading days after the IPO. If an IPO firm is delisted before the end of the 5-year period, returns are compounded until the delisting date. Abnormal holding period returns for each IPO firm are calculated as a difference between its holding period return and a corresponding benchmark's holding period return. Firms with stronger antitakeover protection have four or more firm-level ATPs, and firms with weaker antitakeover protection have three or fewer firm-level ATPs. Significance levels are based on the parametric 1-test for the difference in mean abnormal holding period returns between the two samples and the Wilcoxon rank-sum test for the difference in median abnormal holding period returns between the two samples.

EW = equal weighted; VW = value weighted.

+ Significant at the .10 level.

^{*} Significant at the .05 level.

^{**} Significant at the .01 level.

Relationship among Management Quality, Antitakeover Provisions (ATPs), and Post-Initial Public Offering (IPO) Stock Return Performance for Firms with IPOs 1993-2000 Table 13

		Group A	Group A $(N = 209)$			Group B	Group B $(N = 502)$		Difference in Mean	Difference in Median
		Mean		Median		Mean		Median	Abnormal Holding	
	Mean IPO	Benchmark	Median IPC	Benchmark		Benchmark	Mean IPO Benchmark Median IPO	Benchmark	Period Returns	Period Returns
Benchmark	Return	Return	Return	Return	Return	Return	Return	Return	t-Statistic	z-Statistic
NASDAQ EW Index	23.86	281.25	-50.88	270.55	23.62	340.77	-47.20	311.20	59.76** (2.966)	36.97** (3.552)
NASDAQ VW Index	23.86	87.74	-50.88	65.29	23.62	124.60	-47.20	106.44	37.10^{*} (2.045)	37.48** (3.226)
CRSP EW Index	23.86	173.31	-50.88	175.04	23.62	206.99	-47.20	199.44	33.92^{+} (1.917)	20.72** (3.060)
CRSP VW Index	23.86	67.97	-50.88	54.40	23.62	96.87	-47.20	94.02	29.14^{+} (1.675)	35.94** (3.257)
S&P 500 Index	23.86	64.52	-50.88	53.21	23.62	94.24	-47.20	94.72	29.26^{+} (1.716)	37.83** (3.321)

parametric t-test for the difference in mean abnormal holding period returns between the two samples and the Wilcoxon rank-sum test for the difference in median abnormal holding period returns between the two samples. EW = equal weighted; VW = value weighted.

* Significant at the .10 level. each IPO firm are calculated as a difference between IPO firm's holding period return and a corresponding benchmark's holding period return. Group A represents firms with above median management quality factor scores and four or more firm-level ATPs. Group B represents the rest of the sample. Significance levels are based on the Note. Data are percentages unless otherwise specified. Returns are calculated by compounding daily returns for 1,260 trading days after the IPO. If an IPO firm is delisted before the end of the 5-year period, returns of IPO firms and corresponding benchmarks are compounded until the delisting date. Abnormal holding period returns for

^{*} Significant at the .05 level.

^{**} Significant at the .01 level.

performance than firms in group B and provide support for the long-term value creation hypothesis (hypothesis 5).²⁸

In summary, our results on post-IPO stock return performance provide support for the long-term value creation hypothesis and contradict the managerial entrenchment hypothesis.

6. Conclusion

In this paper we shed new light on the role of ATPs in firms going public by analyzing the relationship between the quality and reputation of a firm's management and the prevalence of ATPs in the corporate charters of IPO firms. Using hand-collected data on the quality of the management teams of firms going public and on the ATPs in their corporate charters, we study the relationship among management quality, ATPs, and post-IPO performance and IPO valuation for the first time in the literature. We test the implications of two theories: the managerial entrenchment hypothesis, which implies that ATPs are meant mainly to enhance the control benefits of existing firm management by minimizing the probability of takeovers by rival management teams and therefore always reduce shareholder value, and the long-term value creation hypothesis, which argues that such provisions, while they entrench firm management, can also be value enhancing in the hands of higher quality managers, since they allow such managers to create superior long-term value for the firm without paying undue attention to short-term pressures arising from the market for corporate control.

Our empirical results are as follows. First, firms with higher quality managers are associated with a greater number of ATPs than are those with lower quality managers. Second, within the high-management-quality category, firms with larger growth options are associated with a greater number of ATPs compared with all other firms. Third, regardless of management quality, firms with a greater number of ATPs outperform those with a smaller number of ATPs, in terms of both post-IPO operating and post-IPO stock return performance. Fourth, if we divide our sample of IPO firms into four categories—those with higher or lower management quality and those with a greater or a smaller number of ATPs within each management quality category—firms with higher management quality and a greater number of ATPs have better post-IPO operating and post-IPO stock return performance and higher IPO valuation than firms in the remaining

²⁸ We also study long-term post-IPO stock return performance by using 5-year wealth relatives for portfolios of IPO firms and corresponding benchmarks. (These results are available on request.) Wealth relatives are calculated as a ratio of the end-of-period wealth from holding a portfolio of IPO firms to the end-of-period wealth from holding a portfolio of a corresponding benchmark. Our findings using wealth relatives are consistent with those reported in Tables 12 and 13. In particular, IPO firms with stronger antitakeover protection have larger wealth relatives across all benchmarks than do IPO firms with weaker antitakeover protection, and IPO firms in group A have consistently larger wealth relatives than do IPO firms in group B across all benchmarks.

three categories. These findings contradict the managerial entrenchment hypothesis and support the long-term value creation hypothesis.

Thus, in contrast to much of the literature on the role of ATPs in corporate charters in the context of IPOs and other corporate events, the evidence presented here indicates that having stronger ATPs in a firm's corporate charter does not necessarily destroy shareholder value. The evidence emerging from our empirical analysis indicates that ATPs play a more nuanced role in affecting shareholder value: while a greater number of ATPs may indeed destroy shareholder value in the hands of poorer quality management (possibly by entrenching it more strongly), our analysis indicates that ATPs are value enhancing in the hands of higher quality firm management. Thus, by incorporating the role of management quality into our empirical analysis, we provide a more complete picture of the relationship between the prevalence of ATPs in a firm's corporate charter and shareholder value.

Appendix A

Descriptions of Antitakeover Provisions (ATPs)

Provision	Description
	Firm-Level ATPs
1. Antigreenmail	Greenmail refers to targeted stock repurchases by management, usually at a substantial premium over market value, of company shares from groups or individuals seeking control of a company; antigreenmail provisions prohibit managers from entering into such arrangements with bidders unless they are approved by shareholders or the same repurchase offer is made to all shareholders
2. Blank-check preferred stock	Preferred stock that is authorized but not issued; it gives a company's board of directors power to issue shares of preferred stock that is authorized but not issued; distribution, conversion, and other rights at the time of the issue; blank-check preferred stock can be placed with friendly parties to deter potential takeover bids by diluting bidders' equity and voting positions it can be also used to establish poison wills.
3. Staggered (classified) board	A board of directors that is usually divided into three classes, with each class elected in different years and serving a 3-year term; classifying the board makes it more difficult to change control of the company through proxy contests since a minority of factors are elected each year; a bidder who has voting control of the company will be unable to gain control of the hoard in a sincle election and would need in to 2 years to gain control.
4. Fair price	Usually adopted to defend against two-tiered front-end-loaded tender offers when the bidder first buys a controlling block of shares and then offers a lower price to remaining shareholders, which usually forces the target shareholders to tender their shares in the first stage regardless of the price offered, since the second-stage price will be lower; fair-price provisions usually require bidders to pay remaining shareholders the same price paid to acquire the controlling block in the first stage; the bidder may avoid such pricing requirements if the offer is approved typically by the supermajority of disinferested characholders or the hoard of directors.
5. Poison pills	Also known as shareholder rights are points are financial instruments in a form of rights or warrants issued to shareholders that trade with common shares; when triggered by a hostile takeover attempt, poison pills detach, trade separately, and become valuable; poison pills can dilute a bidder's equity holdings and voting interests in a target company by giving a right to common shareholders to buy additional shares of the target company at a steep discount or can dilute a bidder's equity holdings in a merged company by giving a right to a target firm's shareholders to buy discounted shares of the postmerger company.

6. Stakeholder clause

Shareholder meeting restrictions: 7. Meetings called only by directors or executives

8. Supermajority required to call special meetings

9. Advance-notice requirement

10. Restrictions on action by written consent

Supermajority vote requirements: 11. Supermajority required to approve mergers

12. Supermajority required to replace directors13. Supermajority required to

13. Supermajority required to amend charter and bylaws14. Unequal voting rights

employees, creditors, suppliers, customers, surrounding communities, and others; provides a target firm's directors with a Permits directors, when evaluating takeover bids, to consider the interests of constituencies other than shareholders such as legal basis to take actions that could be value decreasing to shareholders, for example, turning down attractive takeover

Allows common shareholders to call special shareholder meetings if they can get the consent of a shareholder or groups of Disallows common shareholders and authorizes only directors or executives to call special shareholder meetings to act on matters that arise between regularly scheduled meetings; can deter potential takeovers by delaying removal of directors by a controlling bidder or by hindering the ability of common shareholders to vote on attractive bids shareholders holding a supermajority of outstanding shares

specifies a window for the earliest and latest dates for such submissions, for example, no later than 60 days prior to the ability of shareholders to act by written consent by prohibiting it or requiring unanimous or majority written consent; meeting with a submittal window of at least 60 days, advance-notice requirements can deter takeovers by prohibiting Action taken without a meeting if shareholders individually or collectively consent in writing to such action; limits the Requires shareholders to give advance notice of matters they intend to present at the shareholders' meeting; usually shareholders from voting on matters regarding the takeover bids if proper advance notice is not submitted can delay takeovers by forcing a bidder to take an action at the next scheduled meeting

Requires the vote of a supermajority of shareholders to replace directors and can deter takeovers by limiting the ability of a Requires the vote of a supermajority (usually at least two-thirds and up to 90 percent) of shareholders to approve mergers, Requires the vote of a supermajority of shareholders to amend charter or bylaws and restricts the ability of shareholders to business combinations, or asset sales; supermajority requirements are often unreachable because they exceed the level of shareholder participation at a meeting or because of large insider or employee stock ownership plan shareholdings bidder to remove directors opposing the takeover

Refer to a share structure with more than one class of common shares that have different voting rights, usually insiders of a firm, such as managers and inside directors, hold a class of shares that gives them more than one vote per share, with repeal other ATPs, which are usually proposed as amendments to charter and bylaws the class held by other shareholders having only one vote per share

6		
	Continuod	
	₫	
	٥	
	,	

Provision	Description
Miscellaneous ATPs: 15. Directors can be removed only for cause 16. Merger must be approved by	Limits the ability of potential acquirers to remove directors opposing a takeover Requires the approval of inside directors or directors not related to a potential bidder for a merger to take effect
Instact curceoss 17. Restrictions on transfer of common stock 18. Restrictions on votes each shareholder may cast	Puts various restrictions on transfer of common stock; for example, such a provision may require principal shareholders to offer their shares first to other principal shareholders before selling them Puts various restrictions on the votes each shareholder may cast; for example, shareholders who own more shares than a prespecified threshold may cast only half of their votes
19. Prohibition of cumulative voting for election of directors	Cumulative voting permits shareholders to combine their votes for directors and distribute them among one, a few, or all directors when more than one director is nominated for election; cumulative voting makes it easy for minority shareholders to elect their own representatives and can be particularly important in proxy contests; prohibition of cumulative voting limits the ability of bidders to elect their own representatives to the board of directors
	State ATPs
20. Freeze-out law	Requires a large shareholder who surpasses a certain share ownership threshold (usually from 5 percent up to 25 percent) to wait for a certain period of time (1–5 years) before gaining control of the firm unless the transaction is approved by the board of directors or the majority of shareholders; even after the waiting period, most acquirers are subject to fairprice provisions, the freeze-out law is the only state antitakeover law in Delaware with a trigger threshold of 15 percent and a waiting period of 3 years
21. Control share acquisition law	Requires that if a shareholder acquires shares that increase her holdings above a certain threshold (or a series of thresholds), these shares do not have voting rights unless they are approved by a majority or supermajority of disinterested shareholders

22. Fair-price law	Similar to firm-level fair-price provisions, state-level fair-price provisions regulate the back-end price of two-tier takeover bids, around half of the states having these provisions allow fair-price requirements to be bypassed if approved by a majority of disinterested shareholders, while approximately 40 percent of states require a supermajority of all shares outstanding; some states allow boards to approve a transaction without shareholder approval; fair-price provisions typically require potential bidders to pay the highest market value of stock on a certain date (for example, the date the tender offer commenced or the date it was announced) or the highest price paid by an acquirer prior to becoming an interested shareholder
23. Poison-pill endorsement law	Removes courts from the position of challenging potentially abusive pills, thus making boards more secure in their right to use poison pills
24. Constituency law	Similar to firm-level stakeholder clauses, grants boards the right to consider the interests of other constituencies such as bondholders, employees, creditor, host communities, and even a society as a whole when evaluating takeover bids
25. Cash-out law	Gives disinterested shareholders the right to sell their shares to a large shareholder, after she takes a controlling position, and this controlling shareholder must buy those shares at the highest acquiring price; allows disinterested shareholders to cash out of their position at the expense of the controlling shareholder
26. Antigreenmail law	State-level antigreenmail provisions prohibit repurchasing shares from large shareholders at a premium if those shares were held for less than a prespecified period of time, unless a transaction is approved by a majority of disinterested shareholders or a board of directors; six states adopted this type of law, and one state later repealed it
27. Disgorgement law	Requires that an acquirer disgorge or pay back to the company any profits realized from the sale of company's stock purchased before achieving control status; prevents potential acquirers from buying a large stake, announcing a battle for control, and then selling the stake at a higher price resulting from the prospect of potential acquisition; adopted in Ohio and Pennsylvania

Sources. Descriptions of firm-level and state-level ATPs mostly follow those in Field and Karpoff (2002) and Daines and Klausner (2001) and are supplemented by additional information found in initial public offering prospectuses.

References

- Agrawal, Anup, and Sahiba Chadha. 2005. Corporate Governance and Accounting Scandals. *Journal of Law and Economics* 48:371–406.
- Agrawal, Anup, and Charles R. Knoeber. 1996. Firm Performance and Mechanisms to Control Agency Problems between Managers and Shareholders. *Journal of Financial and Quantitative Analysis* 31:377–97.
- Agrawal, Anup, and Gershon N. Mandelker. 1990. Large Shareholders and the Monitoring of Managers: The Case of Antitakeover Charter Amendments. *Journal of Financial and Quantitative Analysis* 25:143–61.
- Allen, Franklin, and Gerald R. Faulhaber. 1989. Signalling by Underpricing in the IPO Market. *Journal of Financial Economics* 23:303–23.
- Bebchuk, Lucian Arye. 2003. Why Firms Adopt Antitakeover Arrangements. *University of Pennsylvania Law Review* 152:713–53.
- Bernard, Victor L., and Jacob K. Thomas. 1989. Post-earnings Announcement Drift: Delayed Price Response or Risk Premium? *Journal of Accounting Research* 27:1–36.
- Bertrand, Marianne, and Antoinette Schoar. 2003. Managing with Style: The Effect of Managers on Firm Policies. *Quarterly Journal of Economics* 118:1169–1208.
- Bhagat, Sanjai, and Richard H. Jefferis. 1991. Voting Power in the Proxy Process: The Case of Antitakeover Charter Amendments. *Journal of Financial Economics* 30:193–225.
- Borokhovich, Kenneth A., Kelly R. Brunarski, and Robert Parrino. 1997. CEO Contracting and Antitakeover Amendments. *Journal of Finance* 52:1495–1517.
- Borokhovich, Kenneth A., Robert Parrino, and Teresa Trapani. 1996. Outside Directors and CEO Selection. *Journal of Financial and Quantitative Analysis* 31:337–55.
- Brav, Alon, and Paul A. Gompers. 1997. Myth or Reality? The Long-Run Underperformance of Initial Public Offerings: Evidence from Venture and Nonventure Capital-Backed Companies. *Journal of Finance* 52:1791–1821.
- Cary, William. 1969. Corporate Devices Used to Insulate Management from Attack. Antitrust Law Journal 39:318–24.
- Chemmanur, Thomas J. 1993. The Pricing of Initial Public Offerings: A Dynamic Model with Information Production. *Journal of Finance* 48:285–304.
- Chemmanur, Thomas J., and Paolo Fulghieri. 1999. A Theory of the Going-Public Decision. *Review of Financial Studies* 12:249–79.
- Chemmanur, Thomas J., and Yawen Jiao. 2012. Dual Class IPOs. Journal of Banking and Finance 36:305–19.
- Chemmanur, Thomas J., and Imants Paeglis. 2005. Management Quality, Certification, and Initial Public Offerings. *Journal of Financial Economics* 76:331–68.
- Chemmanur, Thomas J., and An Yan. 2004. A Theory of Corporate Spin-Offs. *Journal of Financial Economics* 72:259–90.
- Comment, Robert, and G. William Schwert. 1995. Poison or Placebo? Evidence on the Deterrence and Wealth Effects of Modern Antitakeover Measures. *Journal of Financial Economics* 39:3–43.
- Core, John E., Wayne R. Guay, and Tjomme O. Rusticus. 2006. Does Weak Governance Cause Weak Stock Returns? An Examination of Firm Operating Performance and Investors' Expectations. *Journal of Finance* 61:655–87.
- Cotter, James F., Anil Shivdasani, and Marc Zenner. 1997. Do Independent Directors Enhance Target Shareholder Wealth during Tender Offers? *Journal of Financial Economics* 43:195–218.

- Daines, Robert, and Michael Klausner. 2001. Do IPO Charters Maximize Firm Value? Antitakeover Protection in IPOs. *Journal of Law, Economics, and Organization* 17:83–120.
- D'Aveni, Richard A. 1990. Top Managerial Prestige and Organizational Bankruptcy. *Organization Science* 1:121–42.
- DeAngelo, Harry, and Edward M. Rice. 1983. Antitakeover Charter Amendments and Stockholder Wealth. *Journal of Financial Economics* 11:329–59.
- Field, Laura Casares, and Jonathan M. Karpoff. 2002. Takeover Defenses of IPO Firms. *Journal of Finance* 57:1857–89.
- Foster, George, Chris Olsen, and Terry Shevlin. 1984. Earnings Releases, Anomalies, and the Behavior of Security Returns. *Accounting Review* 59:574–603.
- Garvey, Gerald T., and Gordon Hanka. 1999. Capital Structure and Corporate Control: The Effect of Antitakeover Statutes on Firm Leverage. *Journal of Finance* 54:519–46.
- Gaver, Jennifer J., and Kenneth M. Gaver. 1993. Additional Evidence on the Association between the Investment Opportunity Set and Corporate Financing, Dividend, and Compensation Policies. *Journal of Accounting and Economics* 16:125–60.
- Gompers, Paul A., Joy Ishii, and Andrew Metrick. 2003. Corporate Governance and Equity Prices. *Quarterly Journal of Economics* 118:107–55.
- 2004. Incentives vs. Control: An Analysis of U.S. Dual-Class Companies. Working Paper No. W10240. National Bureau of Economic Research, Cambridge, Mass.
- Grossman, Sanford J., and Oliver D. Hart. 1988. One Share–One Vote and the Market for Corporate Control. *Journal of Financial Economics* 20:175–202.
- Guay, Wayne R. 1999. The Sensitivity of CEO Wealth to Equity Risk: An Analysis of the Magnitude and Determinants. *Journal of Financial Economics* 53:43–71.
- Hambrick, Donald C., and Richard A. D'Aveni. 1992. Top Team Deterioration as Part of the Downward Spiral of Large Corporate Bankruptcies. *Management Science* 38:1445– 66.
- Harman, Harry H. 1976. *Modern Factor Analysis*. 3d ed. Chicago: University of Chicago Press.
- Harris, Milton, and Artur Raviv. 1988. Corporate Governance: Voting Rights and Majority Rules. *Journal of Financial Economics* 20:203–35.
- ——. 1989. The Design of Securities. *Journal of Financial Economics* 24:255–87.
- Hartzell, Jay C., Jarl G. Kallberg, and Crocker H. Liu. 2008. The Role of Corporate Governance in Initial Public Offerings: Evidence from Real Estate Investment Trusts. *Journal of Law and Economics* 51:539–62.
- Hellmann, Thomas, and Manju Puri. 2002. Venture Capital and the Professionalization of Start-up Firms: Empirical Evidence. *Journal of Finance* 57:169–97.
- Jain, Bharat A., and Omesh Kini. 1994. The Post-issue Operating Performance of IPO Firms. *Journal of Finance* 49:1699–1726.
- Jarrell, Gregg A., and Annette B. Poulsen. 1987. Shark Repellents and Stock Prices: The Effects of Antitakeover Amendments since 1980. *Journal of Financial Economics* 19: 127–68.
- Karpoff, Jonathan M., and Paul H. Malatesta. 1989. The Wealth Effects of Second-Generation State Takeover Legislation. *Journal of Financial Economics* 25:291–322.
- Linn, Scott C., and John J. McConnell. 1983. An Empirical Investigation of the Impact of "Antitakeover" Amendments on Common Stock Prices. *Journal of Financial Eco*nomics 11:361–99.

- Loughran, Tim, and Jay R. Ritter. 1995. The New Issue Puzzle. *Journal of Finance* 50:23–51.
- Masulis, Ronald W., Cong Wang, and Fei Xie. 2007. Corporate Governance and Acquirer Returns. *Journal of Finance* 62:1851–89.
- Megginson, William L., and Kathleen A. Weiss. 1991. Venture Capital Certification in Initial Public Offerings. *Journal of Finance* 46:879–903.
- Michaely, Roni, and Wayne H. Shaw. 1994. The Pricing of Initial Public Offerings: Tests of Adverse-Selection and Signaling Theories. *Review of Financial Studies* 7:279–319.
- Mikkelson, Wayne H., M. Megan Partch, and Kshitij Shah. 1997. Ownership and Operating Performance of Companies That Go Public. *Journal of Financial Economics* 44:281–307.
- Qi, Yaxuan, and John Wald. 2008. State Laws and Debt Covenants. *Journal of Law and Economics* 51:179–207.
- Rechner, Paula K., and Dan R. Dalton. 1991. CEO Duality and Organizational Performance: A Longitudinal Analysis. *Strategic Management Journal* 12:155–60.
- Ritter, Jay R. 1984. The "Hot Issue" Market of 1980. Journal of Business 57:215-40.
- ——. 1991. The Long-Run Performance of Initial Public Offerings. *Journal of Finance* 46:3–27.
- Smart, Scott B., and Chad J. Zutter. 2003. Control as a Motivation for Underpricing: A Comparison of Dual and Single-Class IPOs. *Journal of Financial Economics* 69:85–110.
- Stein, Jeremy C. 1988. Takeover Threats and Managerial Myopia. *Journal of Political Economy* 96:61–80.
- Teoh, Siew Hong, Ivo Welch, and T. J. Wong. 1998. Earnings Management and the Long-Run Market Performance of Initial Public Offerings. *Journal of Finance* 53:1935–74.
- Welch, Ivo. 1989. Seasoned Offerings, Imitation Costs, and the Underpricing of Initial Public Offerings. *Journal of Finance* 44:421–49.
- Williamson, Oliver E. 1975. Markets and Hierarchies: Analysis and Antitrust Implications. New York: Free Press.
- Yermack, David. 1997. Good Timing: CEO Stock Option Awards and Company News Announcements. *Journal of Finance* 52:449–76.